ASSESSMENT OF MARKET PARTICIPATION AND MARKETING CHANNEL CHOICE OF SMALLHOLDER MIXED BEAN PRODUCERS IN ZAMBIA

Style Definition: Heading 1: Font: 12 pt

Style Definition: Heading 2

Style Definition: Heading 3: Font: Bold, Font color: Text 1, Space Before: 12 pt, After: 12 pt

Style Definition: Heading 4: Font: Bold, Space Before: 12 pt, After: 12 pt, Line spacing: 1.5 lines

By

Kennedy Chilimboyi

Department of Agricultural Economics and Extension Studies

A Thesis submitted to the University of Zambia in partial fulfillment of the requirements of the degree of Master of Science in Agricultural Economics

UNIVERSITY OF ZAMBIA

LUSAKA

© March 2021.

DECLARATION

I, Kennedy Chilimboyi declare that this thesis:

a) Represents my own work;

b) Has not previously been submitted for a degree at this or any other University; and

c) The work enshrined in the thesis is the product of my own work except where duly acknowledged.

Signed

Date:

© by Kennedy Chilimboyi 2021. All rights reserved

i

CERTIFICATE OF APPROVAL

This dissertation by Kennedy Chilimboyi is approved as fulfilling the requirements for the award of Master of Science in Agricultural Economics degree by the University of Zambia.



ABSTRACT

Food legumes are an important source of food and cash in developing countries such as Zambia. However, producers of food legumes, which includes mixed beans, continue to face market access challenges exacerbated by maize centric policies. This study evaluated market participation and further assessed factors that influence channel choice of markets among smallholder mixed beans farmers in Zambia. The study used secondary cross sectional data from the Rural Agricultural Livelihoods Survey (RALS) conducted by the Indaba Agricultural Policy Research Institute (IAPRI) in 2015. The sample used in the current study consists of 1,326 mixed beans producing households. The study used descriptives to characterize the mixed beans households by market participation and by marketing channel type. To evaluate factors influencing decision and extent of market participation, the study utilized the Heckman Two-Stage Model. The Multinomial Logit Model was used to assess the factors that influence the market channel choice of households.

Factors that positively influenced market participation in the mixed beans market include education level of the household head, quantity of beans harvested, access to price information, being domiciled in Agro-ecological region III and net off-farm income. Being in Agro-ecological region I had a negative influence on market participation. The education level of household head, area planted, and quantity harvested had a positive influence on extent of market participation. The age of household head, adult equivalent, TLU and being in Agro-Ecological Region I negatively influenced the extent of participation. These results suggest policies aimed at capacity building to improve farmers' production, including marketing and business skills, improving bean farmers' access to price information is essential.

Farmers sell mixed beans to four major marketing channels: small-scale traders, other households, retailers/marketeers and other buyers. Choice of the retailer/marketeer market outlet was influenced by the education level of household head, quantity harvested, access to price information, and access to extension services. Distance to the market and being in Agro-Ecological Region I had a negative influence on the choice of retailer/marketeer. Choice of the other households market outlet was influenced by net off-farm income, and being located in Agro-Ecological Region I. Growing other crops (Simpson Index of Diversity), area planted and quantity harvested negatively influenced choice of the other households market outlet. Education level, quantity harvested and net off-farm income had a positive influence on choosing the other buyers outlet. Being in Agro-Ecological Region I had a negative influence of other buyers market outlet. These results suggest policies aimed at improving price information access, extension service delivery system to enhance decision-making, and investment in rural road infrastructure to aid in accessing profitable markets for beans.

Key Terms: Market participation, Extent of Participation, Marketing channel, Smallholder farmers

ACKNOWLEDGMENTS

The development and compilation of this thesis would not have been possible without the help of many people. First and foremost, I thank the Almighty God for his infinite mercies and kindness to have enabled me reach this far. I sincerely thank my main supervisor Prof. Thomson Kalinda and Co-supervisor, Dr. Chewe Nkonde for their invaluable guidance, encouragement and patience throughout the programme. My gratitude too to the supervisory committee in the Department of Agricultural Economics and Extension Studies for their knowledge, encouragement and assistance.

My immense gratitude to the APPSA Project and the World Bank for the financial support rendered during the study. My sincere thanks to Indaba Agricultural Policy Research Institute for allowing me to use their data. I sincerely thank my family for enduring my absence and for the encouragement and support throughout my study.

I am very grateful to my former class mates: Mr. Lazarous Nyirenda, Mr. Moses Tembo, and Ms. Brenda N. Ambukege for their encouragement and immense support during the entire period of study, God bless you all.

TABLE OF CONTENTS DECLARATION.... ACKNOWLEDGMENTSiv LIST OF APPENDICES LIST OF ABBREVIATIONSxi <u>1.4 Research Questions</u><u>5</u> 1.5 Justification6 <u>1.6 Organisation of the Report7</u> 2.4 Factors affecting smallholder farmer market participation and extent of participation......11

v

| / | Formatted | [|
|--------------------|--------------------|----------|
| // | Formatted | |
| | Formatted | |
| // | Field Code Changed | |
| | Field Code Changed | <u> </u> |
| / | Formatted | <u> </u> |
| | Field Code Changed | <u> </u> |
| | Field Code Changed | <u> </u> |
| | Field Code Changed | <u> </u> |
| - | Field Code Changed | <u> </u> |
| | Formatted | <u> </u> |
| | Field Code Changed | <u> </u> |
| $\langle \rangle$ | Field Code Changed | <u> </u> |
| $\langle \rangle$ | Formatted | <u> </u> |
| $\left(\right)$ | Field Code Changed | |
| $\left(\right)$ | Formatted | <u> </u> |
| | Field Code Changed | <u> </u> |
| $\left(\right) $ | Field Code Changed | <u> </u> |
| $\left\ \right\ $ | Field Code Changed | |
| | Deleted: vii | |
| | Formatted | [|
| | Field Code Changed | [|
| | Field Code Changed | [|
| | Deleted: viii | |
| | Formatted | [|
| | Field Code Changed | [|
| | Field Code Changed | [|
| | Deleted: ix | |
| | Formatted | [|
| | Deleted: x | _ |
| | Formatted | [|
| | Field Code Changed | [|
| | Formatted | [|
| | Field Code Changed | [|
| | Field Code Changed | [] |
| / | Formatted | [] |
| / | Formatted | (|
| / | Deleted: 11 | _ |
| | Formatted | l |

| 2.5.1 Studies on factors influencing channel choice of market outlets |
|--|
| 2.6 Conceptual framework |
| 2.6.1 Justification for inclusion of various variables in the Heckman Two-Stage |
| Equation17 |
| 2.6.2 Justification for inclusion of various variables in the MNL model |
| CHAPTER 3 - METHODOLOGY |
| 3.0 Introduction |
| 3.1 Data Type, Source and Data Collection Methods |
| 3.2 Sample selection |
| 3.3 Data Analysis |
| 3.4 Analytical Framework |
| 3.4.1 Factors Influencing Market Participation |
| 3.4.2 Factors influencing channel choice of market outlet |
| CHAPTER 4 – RESULTS AND DISCUSSION |
| 4.0 Introduction |
| 4.1. Characteristics of Sample Households |
| 4.1.2 Characterizing mixed beans households by market channel |
| 4.2 Econometric Model Outputs |
| 4.2.1 Factors influencing market participation |
| 4.2.2 Factors influencing the extent of participation in the mixed bean market47 |
| 4.2.3 Factors affecting the choice of mixed bean marketing channels |
| CHAPTER FIVE - CONCLUSIONS AND RECOMMENDATIONS |
| 5.1 Introduction |
| 5.2 Conclusions |
| 5.3 Recommendations |

| -{ | Formatted: Font: (Default) Times New Roman, 12 pt |
|----|---|
| -{ | Deleted: 17 |
| -{ | Formatted: Font: (Default) Times New Roman, 12 pt |
| | |
| -{ | Field Code Changed |
| - | Field Code Changed |

Formatted: Font: (Default) Times New Roman, 12 pt

Field Code Changed Formatted: Font: (Default) Times New Roman, 12 pt Field Code Changed

| - | Field Code Changed |
|---|---|
| | Field Code Changed |
| | Formatted: Font: (Default) Times New Roman, 12 pt |

| 5.4. Limitations of the study | | |
|-------------------------------|--------|--|
| | | Field Code Changed |
| <u>REFERENCES</u> | 4 | Field Code Changed |
| APPENDICES | 1 | Field Code Changed |
| A | \sim | Field Code Changed |
| | | Deleted: DECLARATION i¶ CERTIFICATE OF APPROVAL ii¶ |

| | CERTIFICATE OF APPROVAL . IN |
|---|---|
| | ABSTRACT iii |
| | ACKNOWLEDGMENTS iv¶ |
| | TABLE OF CONTENTS v¶ |
| | LIST OF TABLES . vii |
| | LIST OF FIGURES viii¶ |
| | LIST OF APPENDICES . ix¶ |
| | LIST OF ABBREVIATIONS x¶ |
| | CHAPTER 1 - INTRODUCTION . 1 |
| | 1.1 Background 1¶ |
| | 1.2 Statement of the Problem . 3¶ |
| | 1.3 Research Objectives 5 |
| | 1.4 Research Questions . 5 |
| | 1.5 Justification 6 |
| | 1.6 Organization of the Report . 7 |
| | CHAPTER 2 - LITERATURE REVIEW 8 |
| | 2.0 Introduction . 8 |
| | 2.1 Definition of Key Terms 8 |
| | 2.2 Economic Approaches to Modelling Market |
| | Participation 9 |
| | 2.3 Economic Approaches to Modelling Market Channel |
| | Choice 10¶ |
| | 2.4 Factors affecting smallholder farmer market |
| | participation and extent of participation 11 |
| | 2.4.1 Studies on factors influencing participation in |
| | markets 11 |
| | 2.5 Eactors affecting channel choice of market outlet 13 |
| | 2.5.1 Studies on factors influencing channel choice of |
| | 2.5.1 Studies on factors influencing channel choice of |
| | 2.6 Concentual framework 15 |
| | 2.6 1 Justification for inclusion of various variables in the |
| | 2.0.1 Justification for inclusion of various variables in the |
| | A C 2 Justification for inclusion of various variables in the |
| | |
| | |
| | CHAPTER 5-METHODOLOGY 2/1 |
| | 3.0 Introduction - 27 |
| | 3.1 Data Type, Source and Data Conection Methods 2/1 |
| | 3.2 Sample selection . 2/1 |
| L | 3.3 Data Analysis 28 |
| 1 | 3.4 Analytical Framework . 28 |
| 1 | 3.4.1 Factors Influencing Market Participation 28 |
| | 3.4.2 Factors influencing channel choice of market |
| | outlet . 33 |
| | CHAPTER 4 – RESULTS AND DISCUSSION 37 |
| | |
| | 4.1. Characteristics of Sample Households 37 |
| | Formatted: Default Paragraph Font, Font, (Default) |
| | +Body (Calibri) 11 pt Not Bold Check spelling and |
| | arammar |
| | grannindi |
| | Formatted |

LIST OF TABLES

| Table 1: Summary of explanatory variables used in the Heckman two-stage model30 |
|--|
| Table 2: Summary of Variables used in the MNL Model |
| Table 1: Household Characteristics of Bean Producers (Continuous Variables) |
| Table 4: Household Characteristics of Bean Producers (Categorical Variables) |
| Table 5: Factors affecting decision to participate in the mixed bean market (Heckman |
| Selection equation results) |
| Table 6: Factors influencing the extent of participation in the mixed bean market |
| (Heckman outcome equation results) |
| Table 7: Marginal effects of factors influencing choice of market channel outlet 50 |

LIST OF FIGURES

| Figure 1: Diagrammatic representation of the Conceptual framework | <u> 16</u> |
|---|------------|
| | |
| Figure 2: Scatterplot of quantity harvested and quantity sold | 42 |
| Figure 3: Most preferred market channel for mixed beans | <u>44</u> |

LIST OF APPENDICES

х

| • • • • • • • • • • • • • • • • • • • | (| Formatted: Normal |
|---|----------|---|
| Appendix 1. Variance Inflation Factor (VIF) results for the Heckman model | | Deleted: ¶ |
| | \frown | Formatted: No underline, Font color: Auto |
| Appendix 2. Breusch-Pagan / Cook-Weisberg test for heteroskedasticity – | | Formatted: Line spacing: Double |
| Heckman Model77 | | Formatted: No underline, Font color: Auto |
| Appendix 3. Heckman regression results of mixed bean market participation | -(| Formatted: No underline, Font color: Auto |
| and intensity of participation | | |
| Appendix 4. Marginal effects for the Heckman selection equation | -(| Formatted: No underline, Font color: Auto |
| Appendix 5. Marginal effects for the Heckman Outcome equation | -(| Formatted: No underline, Font color: Auto |
| Appendix 6: Variance Inflation Factor (VIF) results for the Multinomial | -(| Formatted: No underline, Font color: Auto |
| Logit Model80 | | |
| Appendix 7. Breusch-Pagan / Cook-Weisberg test for heteroscedasticity - MNL81 | -(| Formatted: No underline, Font color: Auto |
| Appendix 8. Coefficient results of the Multinomial Logit Regression | / | Formatted: No underline, Font color: Auto |
| | | Formatted: No underline, Font color: Auto |
| Appendix 9. Marginal Effects Results – Retailer/Marketeer Market Channel | / | Formatted: No underline, Font color: Auto |
| | / / | Formatted: No underline, Font color: Auto |
| Appendix 10. Marginal Effects Results – Other households Market Channel | | Deleted: Appendix 1. Variance Inflation Factor (VIF) results for the Heckman model 77¶ Appendix 2. Breusch-Pagan / Cook-Weisberg test for heteroskedasticity – Heckman Model. 77¶ Appendix 3. Heckman regression results of mixed bean market participation and intensity of participation. 78¶ Appendix 4. Marginal effects for the Heckman selection equation. 79¶ Appendix 5. Variance Inflation Factor (VIF) results for the Multinomial Logit Model 80¶ Appendix 6. Breusch-Pagan / Cook-Weisberg test for heteroscedasticity - MNL.81¶ Appendix 7. Coefficient results of the Multinomial Logit Regression 82¶ Appendix 8. Marginal Effects Results – Retailer/Marketeer Market Channel .83¶ Appendix 9. Marginal Effects Results – Other households Market Channel 83¶ Appendix 10. Marginal Effects Results – Other households Market Channel 83¶ |

Formatted: Indent: Left: 0 cm, Hanging: 0.79 cm

LIST OF ABBREVIATIONS

Deleted: ¶ ¶

| ACIAR | Australian Centre for International Research |
|-------|--|
| AER | Agro-Ecological Region |
| AEU | Adult Equivalent Unit |
| AGRA | Alliance for a Green Revolution in Africa |
| CSO | Central Statistical Office |
| FAO | Food and Agricultural Organization of the United Nations |
| GDP | Gross Domestic Product |
| GRZ | Government of the Republic of Zambia |
| IAPRI | Indaba Agricultural Policy Research Institute |
| MNL | Multinomial Logit Model |
| MoA | Ministry of Agriculture |
| NAP | National Agricultural Policy |
| OLS | Ordinary Least Squares |
| PABRA | Pan-Africa Bean Research Alliance |
| RALS | Rural Agricultural Livelihoods Survey |
| SID | Simpson Index of Diversity |
| SNAP | Second National Agricultural Policy |
| SSA | Sub-Saharan Africa |
| TLU | Tropical Livestock Unit |
| UN | United Nations |
| VIF | Variance Inflation Factor |
| WB | World Bank |
| WFP | World Food Program |
| ZARI | Zambia Agricultural Research Institute |
| | |

CHAPTER 1 - INTRODUCTION

1.1 Background

The agricultural sector is the primary livelihood source of most poor rural households in Zambia and contributes an average of 9.8% to the nation's GDP (Seventh National Development Plan, 2017). The sector supports livelihoods of over 66% of the country's population, making it a critical livelihood and employment source (Tembo and Sitko, 2013). The Zambian government's focus on crop diversification in rural areas is seen as a solution to address undernutrition, food insecurity, and poverty among other challenges (GRZ, 2016). However, in Sub Sahara Africa (SSA), agriculture remains subsistence, characterised by poor infrastructure, limited access to modern agricultural technology, under-investment and weak policy environment (ACIAR 2012). This limits farming households' income levels. Hence, the need for structural transformation from subsistence agriculture to a market-oriented sector, moving from primary-dependent sectors to diversified economic base (AGRA 2016).

It is however, critical to realise the need for farmers to operate within a commercialised setting to gain significant benefits from their ventures, if market provide incentives to increase production and enhance household income generation (Delgado 1999). Markets are therefore critical in the agricultural transformation process (Sigei 2014). However, majority of the smallholder farmers are domiciled in remote areas with poor transport and poor market infrastructure, leading to the high transaction costs. Transaction costs have been noted to be the key reasons for smallholder farmers' failure to participate in the markets (Makhura *et al.*, 2001).

In Zambia, mixed beans (*Phaseolus vulgaris L.*) is among the important food crops being integrated in the crop diversification programme. The bean crop is essentially grown for consumption; hence area cultivated is usually low. The small areas coupled with low yields lead to difficulties for the smallholder farmers to fully participate in the markets. National yields of all crops in Zambia are considered well below global averages (Tembo and Sitko, 2013). Beans are well known for the nutritional benefits they provide. They are a good source of plant protein, folic acid, dietary fibre, iron, and complex carbohydrates (FAO 1999). Economically, beans are a potential income earner as the surplus is sold as

dry grain or processed (canned) forms. In Zambia, beans are considered one of the most profitable crops, and is considered the second most important economic legume crop (Hamazakaza *et al.*, 2014; Muimui 2010; and Sichilima *et al.*, 2016). Beans are also known for their atmospheric nitrogen fixation capabilities; hence, beneficial in the maize-legume rotations and/or intercropping (Siame *et al.*, 1998; Assefa *et al.*, 2017; Cheruiyot *et al.*, 2001; and Salmerón-Miranda 2008).

The Ministry of Agriculture and Central Statistical Office (2018) reports the average yield for mixed beans was estimated at 0.55 MT/Ha in the 2016/17 season and 0.62 MT/Ha in the 2017/18 season, hence an increase in yield rate of 12.71%. However, these yields are still low compared to potential yields that range from 1MT/Ha to 3MT/Ha (Afriseed, 2018). The importance of beans, in terms of income and food, has been growing over the years (Mumui *et al.*, 2016). Although widely grown in several parts of Zambia, Northern Province accounts for a higher proportion of bean growers (56%), followed by Muchinga (13%), Northwestern (9%), Luapala (6%)' and Central (5%). These provinces lie in AER III with high rainfall patterns. The least is Western province accounting for 1% of the total beans production in the country.

Several factors are said to limit bean production in Zambia. According to Muimui (2010) bean producing farmers encounter multiple constraints, including inadequate production resources leading to low production levels. The Second National Agricultural Plan – Implementation Plan (SNAP-IP) for 2016-2020, notes that despite the growth in the sector, agriculture has continued to be hampered by challenges such as low production and productivity, little value addition, over-dependence on rain for crop production, limited capital and market access for smallholder farmers and high post-harvest crop losses. In terms of yield potential, local cultivars of beans are said to have an average of between 0.3 to 0.5 tons per hectare whilst improved ones range between 1 to 3 tons per hectare (MAFF Info Pack 2000, Hamazakaza *et al.*, 2016).

Produce marketing is essential for every farmer that produces a surplus. Ekoja *et al.*, (2018) posit that agricultural produce marketing is a crucial part of a viable farm enterprise in a food production process. Ekoja *et al.*, (2018) further describe agricultural produce marketing as the flow of goods and services from the point of initial farm production to

ultimate consumer's hands. However, the produce should not just flow without any benefit regarding profit accruing to the producer (farmer). Hence, Onyeabor (2009) adds that agricultural produce marketing depicts a process of sellers' demands and motivation to distribute food items unto ultimate consumers at a profit. It is therefore crucial that markets exist to ensure the produce moves from the producer to the end-user.

Amid the low-income levels in rural Zambia, smallholder farmers must engage in agricultural commercialization through increased market participation. Sigei (2014) postulates that markets act as a pivotal point in the agricultural transformation process. Beans production is one avenue that smallholder farmers can earn income in Zambia. Market participation by bean producers is expected to increase farmers' income as they sell more of their produce. This is against the backdrop that smallholder farmers with low level of market participation are characterized by poverty, smaller cultivated fields, and low yields (Mathenge *et al.*, 2010). In a liberalized market economy like Zambia, farmers face challenges of where to sell their produce, hence it is cardinal to have information on available markets. For instance, Tsourgiannisa *et al.*, (2008) point out that marketing channels used when selling the product has a bearing on the profit farmers may make. Hamazakaza *et al.*, (2014) found that farm gate is the most common point of first sale for beans accounting for 65.5 percent of the transactions since most of the households have no market in the villages of their residences. It is believed that buyers usually offer low prices to farmers for the produce in order to make profits after reselling.

Several studies have investigated market participation and channel choice of markets involving several agricultural enterprises to understand what influences the decisions of farmers to participate in a particular market (Awotide *et al.*, 2013; Amao and Egbetokun 2018) or choose a particular market outlet (Tola 2014; Siddique 2018; Mukarumbwa, *et al*, 2018). Such studies have drawn essential recommendations to influence policy and improve the welfare of the producers of the various agricultural commodities.

1.2 Statement of the Problem

Monoculture associated with maize growing in Zambia has long been recognized and understood to be a threat to income, food and nutrition security of smallholder farmers. Currently, maize dominates Zambian agriculture, covering two-thirds of the cropped area and grown by approximately 98.7% of farming households. The nation's diet mirrors maize dominance and the low diversity of crops, with 63% of total dietary energy derived from cereals (Mwanamwenge and Harris, 2017). Consequently, the Zambian food system is not delivering enough affordable or nutritious foods for most of the population. The 2018 State of Food Insecurity and Nutrition in the World report lists Zambia among the countries with the highest undernourishment rates (FAO *et al.*, 2018). Similarly, Zambia's hunger index remains among the highest globally with 38% of the population estimated to experience hunger (WHH & CWW, 2018).

Amid hunger, poverty, and health concerns, the Government of the Republic of Zambia through its agricultural policy reform updates, has emphasized crop diversification to integrate food legumes in the smallholder farming systems. Mixed beans (*Phaseolus vulgaris L.*) are among other food legume crops that have so far shown promise to contribute to income food and nutrition security of rural households, most of whom live in poverty. Mango, *et al.*, (2018) highlight market participation and marketing channels to significantly affect production and productivity because of their potential to cause inefficiencies in market linkages. This may lead to farmers producing less and selling to limited market outlets, offering low prices. Hamazakaza *et al.*, (2014) found that farm gate is the most common point of first sale accounting for 65.5 percent of the transactions since most of the households have no market in the villages of their residences. Furthermore, Muimui (2010) pointed out inadequate linkages among supply chain actors especially between smallholder farmers and canning industries.

Several studies have been carried out on market participation and channel choice decisions globally to date. These studies have been important in highlighting socio economic and institutional factors influencing market participation (Chalwe 2011; Moono 2015) and channel choice decisions (Mmbando 2014; Tumukunde 2018).

However, despite a wealth of knowledge on market participation, information on factors influencing extent of participation, particularly in the bean sector is limited in Zambia. Chalwe (2011)'s study only focused on factors influencing market participation in the bean market. In addition, information on factors influencing market channel choice of

several bean market outlets is limited in Zambia despite the liberalised market system. The study by Chalwe (2011)' only explored factors influencing marketing channels on two outlets. This study therefore, extends the earlier study by including factors that influence extent of participation, and highlights factors that influence channel choice of multiple outlets in Zambi

1.3 Research Objectives

1.3.1 Overall Objective

This study's objective was to assess market participation and factors that influence extent of market participation and to further assess factors that influence the market channel choice among smallholder mixed beans farmers in Zambia.

1.3.2 Specific Objectives

The specific objectives of the study were to:

- i. To characterize smallholder mixed bean producers by market participation and marketing channel.
- ii. To evaluate the factors that influence the decision and extent of participation by smallholder farmers in the market of mixed beans.
- To assess the factors that influence the choice of marketing channel by smallholder mixed bean producers

1.4 Research Questions

- **i.** What are the characteristics of smallholder mixed bean producers by market participation and type of marketing channel?
- <u>ii.</u> What factors influence smallholder farmers' participation and level of participation in mixed bean markets in Zambia?
- iii. What factors influence the choice of market outlet for mixed beans in Zambia?

Deleted: i.

Formatted: Font: (Default) Times New Roman, 12 pt, Font color: Text 1

Formatted: List Paragraph, Numbered + Level: 1 + Numbering Style: i, ii, iii, ... + Start at: 1 + Alignment: Right + Aligned at: 0.63 cm + Indent at: 1.27 cm

1.5 Justification

Literature contends that market participation by smallholder farmers in developing countries is meagre, a development which has slowed down agriculture driven economic growth and exacerbated poverty levels. For example, in a survey of the literature of smallholder participation in food grain markets covering six countries in east and southern Africa, Barrett (2008) finds that the per cent of net-sellers ranged between 25 per cent (Ethiopia, 1996) to 39 per cent (Kenya, 1998). Mather *et al.*, (2013) also find that the percentage of net-sellers of food grains in Mozambique (2005) and Zambia (2008) were as low as 15 and 27 per cent, respectively. However, increased participation of smallholder farmers in agricultural output markets holds the potential to lift farmers out of high-risk and low-productivity subsistence farming to more commercial and profitable agriculture (Heltberg and Tarp 2002; Barrett 2008; Von Braun *et al.*, 1994; Timmer 1988). This information clearly shows that a relatively low portion of smallholder farmers participate in food markets as net sellers in many sub Saharan African (SSA) countries.

Another less explored aspect of smallholder market participation in the developing country context is the choice of marketing channel that households make when faced with several buyer types, such as, private traders of various scales, government agencies, and other households in the village. According to Tsourgiannis *et al.*, (2008), the marketing channel used when selling the product has a bearing on the profit farmers may make.

Given an understanding that most studies in the literature on the choice of a marketing channel for agricultural output are limited in their analysis to the choice between selling at a distant market versus at the farm gate as well as abstracting away from the potential constraint of imperfect input markets, this study sought to understand farmer circumstances by conducting a wholesome analysis of the technical, environmental and socio-economic factors. The results can be used by policymakers, extension service providers and smallholder farmers to inform decision making.

1.6 Organisation of the Report

The remaining parts of the report are structured as follows. Chapter 2 presents the literature review beginning with the definition of terms and reviews previous research studies on market participation and channel choice of market outlets. Chapter 3, describes the research methodology, theoretical and analytical frameworks, whilst chapter 4 discusses the results of the study, both descriptive and econometric. Chapter 5 presents the conclusion and recommendations based on the findings of this study.

Deleted: z

CHAPTER 2 - LITERATURE REVIEW

2.0 Introduction

This chapter begins with a theoretical literature review on factors influencing market participation and channel choice of market outlet. The chapter reviews selected empirical studies carried out on market participation. The chapter further looks at empirical studies on factors influencing channel choice of various agricultural enterprises' market outlets.

2.1 Definition of Key Terms

Market Participation

According to Otekunrin *et al.* (2019), market participation is one of the key concepts in agribusiness management and has been defined differently by various authors. Generally, market participation can be referred to as participation in market related activity, which results in the sale of produce involving cash or in kind (Otekunrin *et al.* (2019). In an agricultural market economy, market participation can also be referred to as market earnings from market activities (Makhura *et al.*, 2001). In this study, market participation is considered from the perspective of farmers selling their produce in the beans market to earn an income.

Extent of Market Participation

The intensity of market participation is considered from volumes of output sold. Jagwe (2011) defines intensity of market participation as the quantity of output sold by a farmer from the total harvested. In this study, the extent of market participation is considered from the view of quantity of mixed beans sold by farmers in the beans market.

Marketing Channel Choice

Marketing channels are an essential institution of any producer's activities. Han (2011), therefore, defines a marketing channel as the specific organizations that are interdependent and interrelated with products and their relevant services that can be transferred from producers to consumers or sellers. Similarly, Giles (1973), defines marketing channels as the system of marketing institutions through which goods or services are transferred from the original producers to the ultimate users or consumers. According to Mmbando (2014),

most producers do not sell their goods directly to the final users; between them stands a set of intermediaries performing a variety of functions. These intermediaries constitute a marketing channel. The choice of a particular marketing channel is critical and dependent on several factors.

Smallholder Farmer

The definition of a smallholder farmer differs depending on the context, country, agroecological zone and intensity of farming (Vermuelen and Cotula 2010; Nyambo *et al.*, 2019). According to Mmbando (2014), this explains the interchangeable term 'smallholder' with 'small-scale', resource poor' and 'peasant farmer'. Current definitions, however, are centred on common characteristics of smallholder farmers such as access to land, land size, labour, resource endowment, technology and market orientation (Kamara *et al.* 2019). The World Bank (2003) defines smallholders as farmers with a low asset base, operating less than 2 hectares of land of cropland and depending on household members for most of the labour. In Zambia, smallholder farmers are the vast majority among the three farmer categories (small, medium and large-scale). They are characterised by cultivating less than 5 ha, producing mostly for consumption, using few inputs, and selling a small portion of their surplus (Mukutu et al., 2004; IAPRI 2015; Siegel & Alwang 2005;).

2.2 Economic Approaches to Modelling Market Participation

Econometric analysis is used to estimate causal relationships between the dependent and explanatory variables (Moono 2015). In this context, econometric analysis helps to understand the effects of the different factors on market participation and the extent of participation in the beans value chain. Several models have been used to study market participation of agricultural enterprises. These models include the Heckman two-stage, Double-hurdle, Triple-hurdle and Tobit models (Komarek 2010; Moono 2015). The Tobit model (Tobin 1958) was popular, but due to its limitations, the Heckman became more preferred (Sigei 2013). The Tobit model's major limitation is that it assumes that the decision to sell and determine sales volume are influenced by the same set of parameters and variables (Reyes *et al.*, 2012). Hence, the Tobit is more appropriate when the decisions to sell and volume to be sold are simultaneously made.

In contrast, Barrett (2007) notes that households face a two-step decision making process regarding market participation. Firstly, households decide whether or not to participate in the market. After that, the household decides on the quantity to sell. The suitable models in cases where decisions are not simultaneously made include the Double-huddle and the Heckman two-step (Mather *et al.*, 2011). In this study, the Heckman two-step procedure is used to cater for sample selection bias arising from the data used. The Heckman two-stage model developed by Heckman (1979) has been used to correct for biases arising from sample selection using simple regression. Heckman (1979) demonstrated that the bias that results from using selected samples could be due to omitted variables. As a solution, he proposed that, for the full sample (e.g. market participants and non-market participants), a probit analysis could be used to estimate the individual's probability to be in the selected sample (e.g. will participate in the market). Furthermore, Barret (2007) demonstrated that by using this probability as a regressor in the equation of interest (e.g. farmer's yield), unbiased estimators could be obtained to estimate the extent of participation.

2.3 Economic Approaches to Modelling Market Channel Choice

The Multinomial Logit Model (MNL) was first introduced by McFadden (1974) to explain the choice of transportation modes of urban commuters with the random utility model. In this study, the MNL is used to explain the effects of the independent variables on the preference of a specific market channel for mixed beans. The MNL continues to be a preferred model since it allows the analysis of decisions across more than two categories in the dependent variable (Woodridge, 2008). The MNL model, therefore, makes it possible to determine the choice probabilities of multiple channels. Dougherty (1992) points out that the procedure for formulating a multinomial logistic regression is the same as for binary logistic regression. However, the MNL model differs from the binary Probit and Logit models, which are limited to the maximum of two choice categories (Maddala, 1983). A binary choice model is used where the outcome being investigated is binary in its estimation (Greene, 1997). Hence, the MNL Model choice in this study was based on choice probabilities of multiple channels. The MNL also suffices when dealing with mutually exclusive dependent variable. In this study, the market channels are considered mutually exclusive. The assumption is that each farmer faces a set of discrete, mutually exclusive choices of market channels. This means that a farmer only chooses one channel at a time to sell their beans and not both/all channels. If the dependent variable is categorical and mutually exclusive, then the dependent variable is assumed to have a multinomial distribution (Green, 1997).

2.4 Factors affecting smallholder farmer market participation and extent of participation

Several factors are known to affect smallholder farmer participation in the markets. These factors range from socio-economic, institutional, market and external factors (Goetz 1992; Sigei 2014; Apind 2015). These factors can negatively or positively influence market participation and the extent of participation of smallholder farmers. Social-economic factors include age, gender, education level, and farming experience of the household head, household size, farm size, level of output, and ownership of productive assets (Sigei 2014; Apind 2015). Institutional factors include: access to extension service, access to credit, membership to groups, and infrastructure. Market factors that may influence market participation include: access to market information, price and access to price information, distance to markets, and transport means. Political stability and natural phenomena such as drought or floods affecting a nation are some of the external factors that may influence market participation (Goetz 1992; Sigei 2014; Apind 2015).

2.4.1 Studies on factors influencing participation in markets

Several studies have been carried out to determine factors influencing market participation and the extent of participation in Sub Sahara Africa and other parts of the world.

Masuku *et al.*, (2001) assessed the factors influencing the decision to sell maize and the choice of marketing chain by smallholder farmers in Swaziland. The study employed a logistic regression on the smallholders' decision either to sell or not to sell maize. Factors such as off-farm income activities, experience, access to agricultural information, participation in agricultural schemes, family members without education, and farm size were found to influence the decision to participate in the maize market. Transportation

costs and farm size were found to influence the choice of the maize marketing channel. The study is similar to Chalwe 2011 in that it only focused on assessing factors influencing the decision to participate in the market.

A study by Mathenge et al., (2010) examined the factors influencing market participation of crop producers and their impacts on income and poverty among the poor and marginalized groups in Kenya. The study employed the Heckaman two-stage model, and modelled participation in input and output markets. The study focused only on poor and marginalized farmers. Takele (2010), also used the Heckman two-step model to examined the profitability and marketing chain of rice in Fogera Woreda, South Gondar zone of Amhara Regional State in Ethiopia. The Heckman's two-step selection was used to examine determinants of household's rice supply to markets. Takele (2010) further employed the Tobit to answer questions on factors that affect market participation and extent of participation at the same time. Using the Tobit model, the study assumed that all producers are likely suppliers of rice, and that same variables equally influence market participation and extent of market participation. However, Reyes et al., (2012), used a Double hurdle model to estimate the factors influencing marketing decisions among smallholder potato growers in the central highlands of Angola using cross-sectional data from three provinces of Angola. The Double hurdle model was used because it is a more flexible alternative than the Tobit model since it allows for the probability that factors influencing the decision to sell a crop be different from factors affecting the decision of how much to sell.

Other studies (Awotide *et al.*, 2013; Tesfaw 2013), used the Heckman two-step model to identify the determinants of market participation in the rice and pepper markets. Kuma *et al.*, (2014) focused on assessing factors affecting milk market participation and volume of supply in volume in Wolaita Zone, Ethiopia. Since, Kuma *et al.*, (2014) focused on a perishable agricultural commodity (milk). Factors influencing participation would therefore be expected to be different with the other studies on non-perishable commodities (rice and pepper).

Studies by Mathenge *et al.*, (2001), Takele (2010), Tesfew (2013), Awotide (2013), and Kuma (2014) were very beneficial to the current study in providing insights into the

various socio-economic, farm level, and market and institutional level variables to use in the model. Even though, they analyzed different agricultural produce, the studies employed the Heckman two-step model, which the current study used to determine factors influencing market participation and extent of market participation for mixed beans farmers. Chalwe (2011)'s study on factors influencing market participation in the beans market in Zambia, used the probit model. However, the current study used the Heckman two-step model, which extends Chalwe (2011) study by including an assessment of factors influencing extent of participation in Zambia.

2.5 Factors affecting channel choice of market outlet

The farmers' decision on where to sell or not to sell their farm produce is referred to as 'channel choice of market outlet'. According to Qadri (2018), channels of distribution are the ways and means employed by manufacturers and sellers to get their products to the market and into the users' hands. In this study, the market channel refers to an outlet where mixed beans produced by farmers ends up. The mixed beans producer chooses among the available market channels for beans. According to Sigei (2014), market outlet choice is determined by the price the farmers receive from the sale of produce. Hence, there is a higher likelihood of the farmer to choose the channel, which is more lucrative. However, various factors are expected to influence the farmers' channel choice decision. The factors that influence channel choice of market outlet include, age and farming experience of household head, farm size, access to extension services, membership to a group(s), the area planted, price and price information, geographical location, and distance to market (Sigei 2014).

2.5.1 Studies on factors influencing channel choice of market outlets

The MNL model has been used in various studies to determine the factors influencing the channel choice of marketing outlets using the MNL model. The advantage of using the MNL model is that it can be applied on multiple-choice categories, unlike the probit or logit models, which are limited to a maximum of two choice categories (Maddala 1983; Wooldridge 2002). The current study is based on multiple choice categories.

Ferto and Szabo (2002) employed a multinomial logit model to identify determinants influencing producers' decision for channel choices of fruit and vegetable producers in Csongrad, Hungary. The study identified three supply channels for namely wholesalers, marketing cooperative and producers' organization. However, a study by Ogunleye and Oladegi (2007) analyzed determinants of the choice of marketing channel choice among cocoa farmers in Nigeria using descriptive analysis. The study by Ferto and Szabo (2002) differs from Ogunleye and Oladegi (2007)'s study differs from Ferto and Szabo (2002)study in that quantitative analysis was used to determine factors influencing the choice of marketing channels in the latter, whilst only descriptives were used in the former.

Chalwe (2011) carried out a study on factors influencing bean producers' choice of marketing channels in Zambia and used the multinomial probit model, focusing on two market outlets. However, most studies (Mmbando *et al.*, 2016; Maina, *et al.*, 2015; Singh 2018) used the MNL to study factors influencing choice in multiple marketing channel outlets. Similarly, Musara *et al.*, (2018) examined the determinants marketing channel choice decisions by small scale sorghum farmers in Zimbabwe's Mid Zambezi valley by segmenting three common marketing channels for sorghum (local, market and combination).

Studies by Ferto and Szabo (2002), Maina *et al.*, (2015), Mmbando *et al.*, (2016), Singh (2018), Musara *et al.*, (2018), and Mukarumbwa *et al.*, (2018) were very beneficial to the current study in providing insights into the various socio-economic, farm level, and market and institutional variables to use. Even though they analyzed different channels, the studies all used the MNL model. The MNL model is considered the best approach for choices based on the decision-maker attributes than the choice itself. Despite Chalwe (2011)'s study, factors influencing choice of beans marketing channels have not been extensively investigated in Zambia. Chalwe (2011) study only focused on two market outlets. In reality more than two channels for marketing products exist. Furthermore, several studies carried out on channel choice decisions have commonly adopted the MNL as a more appropriate method for analyzing factors affecting choice of marketing channels, which the current study endevoured to use.

2.6 Conceptual framework

Figure 2 shows the interrelationships in the study, including interrelations of key variables. The characteristics of interest include socio-economic, farm level, and market and institutional factors. Socio-economic characteristics include factors such as household head's age, sex, and education, family size, and off-farm income. Farm level factors include landholding size, area planted, quantity harvested, quantity sold, the price per kg, and production assets owned. Institutional and market factors include price information, cooperative membership, extension services, and distance to market. Location factors include geographical locations, which capture the pros and cons of different locations with regard to market participation, extent of participation and channel choice of market outlet. The factors mentioned above influence market participation in the mixed beans market, including the extent of participation. The participation in the market and extent of participation in the market, coupled with extent of participation and selection of lucrative market channels, contribute to agricultural commercialization and increased income.



Figure 1: Diagrammatic representation of the Conceptual framework

Source: Own conceptualization based on literature

2.6.1 Justification for inclusion of various variables in the Heckman Two-Stage Equation

Socio-economic Factors

Age of household head: The smallholder farmer's age is a continuous variable measured in years and was used as a proxy for experience in production and marketing. Age is expected to either have a negative or positive influence on the decision to participate in the mixed beans market or not. Regarding the negative influence, it is expected that younger farmers are more energetic and enthusiastic and therefore can participate in marketing of beans than older farmers (). However, the older farmers are expected to use the experience to increase production and to locate better markets for their produce. A study by Chalwe (2011) found age to have a negative and significant influence in the decision to participate in the beans market.

Adult Equivalent Index: is a continuous variable that is a proxy for available household labour force in adult equivalents. A higher adult equivalent is expected to positively associate with participation in the mixed bean market. Mmbando *et al.*, (2016) contends that since the agricultural activities are labour intensive, households with a larger labour force can cultivate larger areas of land and produce more surpluses to market.

Maximum level of Education: is a continuous variable that indicates the number of years in formal school for the household head. Education is associated with efficiency and a better understanding of production and marketing strategies. Hence an educated beans farmer is likely to produce surplus and participate in marketing. Studies by Sigei (2014) and Astewel (2010) found that education level positively influenced market participation in pineapple and paddy rice, respectively.

Net off-farm income: is a continuous variable measured in terms of income the producer earned from off farm activities and is expressed in Zambian Kwacha. Off farm income is assumed to be a competitor with on-farm income. This is because the more the farmer earns from off-farm income, the lesser the likelihood of producing food crops such as mixed beans. Hence, off-farm income is expected to have a negative relationship with market participation. Mbitsemunda and Karangwa (2017) found a negative relationship between off-farm income and the decision to participate in Rwanda's bean market.

Livestock ownership: Is an index that represents ownership of livestock by the household. It is envisaged that a farmer who owns more livestock is wealthy and can invest adequately in the crop of interest. Therefore, livestock ownership is hypothesised to positively influence market participation in the mixed beans market. Tekele (2010) found that the total value of livestock owned increased the probability of participating in the rice market.

Farm Level Factors

Total landholding size: This variable could influence both decisions in market participation and the extent of participation decisions. It is a continuous variable denoting the total area owned by the farmer expressed in hectares. It is envisaged that a larger area owned by the household implies more area allocated to the crop of interest. Hence, increasing the area allocated to beans will directly lead to surplus and subsequent market participation. According to FAO (2015), the larger the farm, the larger the production proportion sold in the markets. This is because larger volumes are assumed to lower transaction costs. Abayneh and Tefera (2013) found farm size to have a positive and significant effect on the decision to participate in the haricot bean market. Equally, Amao and Egbetokun (2018) found that farm size positively influenced the quantity of leafy vegetables supplied to the market.

Area planted: It is a continuous variable that refers to total area allocated to mixed beans by the farmer. The probability to participate in the beans market is expected to increase when area planted is increased. Similar to landholding size, it is envisaged that a farmer who increases area planted could subsequently have surplus production. The variable is therefore expected to have a positive influence on the extent of market participation of the farmer.

Total quantity harvested: is a continuous variable measured in kilogrammes. Quantity harvested is believed to positively influence the decision to participate in the mixed beans

market. An increase in mixed beans harvested is expected to increase the probability and subsequent proportion that the farmer sells. Mbitsemunda and Karangwa (2017) found that the quantity produced positively influenced the household's decision to participate in the bean market.

Simpson Index of Diversity: This index is used to measure the degree of crop diversity. A score closer to 1 indicates increased diversity in terms of number of crops grown. Farmers who grow more crops are expected to participate in the mixed beans market. Growing more crops is also expected to influence the extent of participation. This is because growing diverse crops may increase the farmer's income sources. Rhaman *et al.*, (2009) noted that crop diversification is an important factor of income diversity. More so, growing various crops may enable the farmer to only sell the most profitable crop leaving the others for consumption. Therefore, SID-crops is expected to have a positive influence on market participation. However, a study by Tesfaw (2013) found that production of other crops together with pepper has a negative influence on extent of participation. This is attributed to a lack of specialization in one crop

Institutional Factors

Access to price information: is a dummy variable taking a value of one if the farmer has access to market price information and value of zero if otherwise. It is assumed that access to price information helps the farmer make informed decisions on how much and where to sell to. Therefore, price information is hypothesised to have a positive influence on market participation. Sigei (2014) found that price information positively influenced the extent of market participation in the pineapple market.

Distance to market: is a continuous variable that is measured in kilometres. The variable represents the distance from the producer's residence to the point of sale (market). The shorter the distance, the lesser the marketing costs that are likely to be incurred by the producer. It is therefore, expected that distance will have a negative influence on marketing participation. However, distance to the selling point may have a positive influence depending on the farmers's preference. Zamasiya *et al.*, (2014) noted that smallholder farmers tend to opt for distant markets if local markets are not offering

lucrative returns. A study done by Abayneh and Tefera (2013) found a negative relationship between distance to the market and participation in the haricot bean market.

Access to loans: is a dummy variable that representing the farmer's access to credit facility for farming purposes from formal and informal sources. If the farmer has access to loans, the value taken is one, and zero if otherwise. It is expected that credit will have a positive influence on market participation. This is because credit is known to boost agricultural production as it enables farmers to purchase the needed inputs, implements and labour. Abayneh and Tefera (2013) and Mbitsemunda and Karangwa (2017) found credit to have a positive and significant relationship with market participation in the bean market. Furthermore, Muhammed (2011) found a positive relationship between credit and the extent of participation in the pepper and teff market.

Cooperative membership: Is a dummy variable taking a value of one if the household member belongs to a cooperative and value zero if otherwise. Farmer groups or cooperatives have been found to increase access to inputs and relevant production and market information. Hence, membership tin a cooperative is hypothesised hypothesised to positively influence market participation of the beans producer. Mathenge *et al.*, (2010) found that belonging to a farmer organization had positively influenced market participation in Kenya's agricultural markets.

Access to extension services: It is a dummy variable taking a value of one if the farmer has access to extension services and zero if otherwise. The variable assesses whether or not the farmer has access to crop diversification and other related information. Access to such information encourages farmers to venture into diverse crops for consumption and selling purposes. Hence, access to extension services can boost production and in turn, positively influence market participation. Siziba *et al.*, (2011) observed that extension training and participation in research positively influenced cereal producers' market participation In addition, a study by Tekele (2010) found that access to extension service increases the household's probability to participate in the rice market.

Location Factors

The Agro-Ecological Regions (AER) refer to the households' geographical locations and are to capture advantages and disadvantages of the different locations with regard to market participation. The Agro-Ecological regions are mutually exclusive; hence, the farmer is only expected to be found in one region and not in two or more at the same time. The relationships revealed by the results are to be explained by the specific attributes of each of the locations. The differences in rainfall pattern in these areas determine the mixed beans production levels and channel choice of market outlets. Three dummy variables, one for Agro-Ecological Region I (AER I), one for Agro-Ecological Region II (AER II) and one for Agro-Ecological Region III (AER III) were defined. Agro-Ecological Region I (AER I) was coded as a dummy variable taking a value of one if the household is found in Agro-Ecological Region I and the value zero if otherwise. AER I is characterized by low rainfall (below 1000mm) and covers Zambia's Southern and Western parts of Zambia. Because of the low rainfall found in this region, the production of mixed beans is equally low. Therefore, households found in this region are likely to produce smaller quantities of beans, hence are not likely to participate in the market. Agro-EcologicalRegion III (AER III), was coded as a dummy variable taking a value of one if the household is found in Agro-EcologicalRegion III and the value zero if otherwise. AER III is characterised by high rainfall per annum (1200mm and above) and lies in the country's Northern and North-Western parts, favouring beans production (Hamazakaza et al., 2016). Unlike farmers in AER I, farmers in this region are likely to produce higher quantities of mixed beans. The larger quantities produced are subsequently expected to positively influence market participation and the extent of participation in the mixed beans market.

2.6.2 Justification for inclusion of various variables in the MNL model

The independent variables with the potential to influence market channel choice decisions adopted from literature were explored.

Formatted: Heading 3, Left, Space Before: 0 pt, After: 0 pt, Line spacing: single Formatted: Font: Not Bold

Socio-economic Factors

Age of household head: The smallholder farmer' age is a continuous variable measured in years, showing how old the household head was. The variable was used as a proxy indicator for experience in production and marketing beans. The aged household heads are believed to be experienced in looking for markets that offer higher prices. This belief is in line with Tumukunde (2018), who posits that the older the household head, the more information he/she has about profitable marketing channels as they have a network with many actors of the value chain. However, Mukarumbwa *et* al., (2018) argue that younger farmers are expected to be risk-takers, innovative and to be involved in activities such as value addition so as to access alternative markets. Therefore, age is expected to positively or negatively influence the channel choice of mixed beans outlet.

Gender of household head: Is a dummy variable taking the value of one if the farmer is male and the value of zero if otherwise. Male headed households are considered to have more access to resources and market information than their female counterparts. These attributes can enhance the male headed households' capacity to choose lucrative markets for their produce. Besides, Kihoro (2016), posits that men have a higher decision making capability within the household shaped by the norms and roles set out for men in the African culture. Hence the gender of the household is likely to influence the channel choice outlet for mixed beans.

Decision maker: is a dummy variable taking the value of 1 if the decision maker is female and the value of zero if otherwise. The decision-maker variable captures essential information on decision making between men and women in bean marketing in Zambia. Women in Africa provide most of the labour and manage many farming activities daily (FAO, 1994; ILO, 2009). It is also known that women and men grow legume crops, but men tend to take over the decision-making if a crop becomes profitable. This, therefore, tends to limit women from reaching their full potential.

Adult Equivalent Index: is a continuous variable that is a proxy for household labour force. A high labour endowment is expected to contribute to the production and marketing of relatively larger quantities, which can then be transported to the retail market (Mabuza

et al., 2014). This entails that the household is likely to sell to the retailer at a market, offering a higher price than the buyer at farm-gate level.

Maximum level of Education: is a continuous variable that indicates the number of years in formal school for the household head, who is the decision-maker. Education level of the decision-maker is associated with efficiency and better understanding of marketing strategies. Some market channels are complicated, hence require skills and understanding obtained through education of the decision-maker. Anteneh *et al.*, (2011) found that household head's the level of education significantly influenced the coffee market outlet's choice.

Net off-farm income. This is a continuous variable denoting the household's net income earned by the household from off-farm activities expressed in Zambian Kwacha. It is expected that off-farm activities add income to the household, thereby enhancing the households's ability to sell to alternative markets away from the farm. Shiferaw *et al.*, (2006) posit that wealthier farmers were likely to be affected by the delay of payment, and therefore, they would choose the channel that provided high prices.

Tropical Livestock Units: is an index used as a proxy for livestock numbers owned by the household. The higher the index, the higher the number of livestock owned. Farmers with more livestock are considered to be wealthy, which means they can allocate adequate resources into beans production to increase yields. In Ethiopia, Hailu and Fana (2017) reveal that TLU had a significant influence on the choice of market outlet for major vegetable crops among smallholder farmers of Ambo and Toke-Kutaye districts. Households with more livestock tend to allocate more of their land to grazing land or fodder production. This entails a reduction in land used for vegetable production and subsequent low production and limited surplus for sale. With this background, the households opt to sell to the direct market outlet relative to the wholesalers who demand for larger quantities.

Farm level Factors

Landholding size: is a continuous variable denoting the total area owned by the farmer expressed in hectares. It is envisaged that larger area owned by the household implies more area allocated to the crop/s of interest. It is, therefore, hypothesised that increasing the area allocated to beans will directly lead to surplus production and subsequent market participation. Singh (2018) argues that households with larger landholdings have less probability of participation in an informal market channel and more in a formal one. This entails that landholding size is an influencing factor in channel choice decision.

Area planted: The area planted is also a continuous variable that measures area planted to beans and is expressed in hectares. Similar to total landholding size, the more the area planted the greater the expected harvest. Therefore, it is envisaged that a larger area planted positively influences the choice of market channel for mixed beans. This is because the quantities that are harvested determine where to sell. Some buyers prefer to purchase in bulk, whilst others prefer smaller quantities.

Quantity harvested: is a continuous variable measured in kilogrammes and shows quantity of beans harvested. A marginal increase in mixed beans harvested has a definite and significant influence on marketable surplus and choice of a particular market channel. Chalwe (2011) found beans yield to influence the choice of selling to private traders than to other households at farm-gate. In another study, Mukarumbwe *et al.*, (2018) noted that the more the quantity of vegetables produced, the more likely the chances that the farmer would participate in distant urban markets.

Simpson Index of Diversity: This index is used to measure the degree of crop diversity. A score closer to 1 indicates increased diversity in terms of the number of crops grown. Growing diverse crops allows the farmer to reserve the crop or crops that give them higher income for sale. The other crops can then be used for consumption. This entails that the crops reserved for sale can then be sold to bulk buyers compared to households who purchase small quantities. Hence, SID-Crops is expected to influence the selection of a market outlet for mixed beans.
Institutional Factors

Access to price information: Is a dummy variable taking a value of one if the farmer has access to market price information and a zero value of otherwise. It is assumed that access to price information is cardinal as it helps the farmer make informed decisions on which market channel offers the best price for their produce. Sigei (2014) found that price information positively influenced the choice of a market outlet in the pineapple marketing.

Distance to market: It is a continuous variable that is measured in kilometres. The variable represents the distance from the producer's production area to the nearest point of sale and can be used as a proxy for transaction costs. Mukurumbwa *et al.*, (2018) point out that the farmer's profit returns are bound to be reduced due to increased transportation costs and the encountered opportunity cost of time. Riziki *et al.*, (2015) found that distance to the market is a significant determinant for channel choice outlets for the African indigenous vegetables in Kenya

Access to extension services: It is a dummy variable taking a value of one if the farmer has access to extension services and zero if otherwise. The variable assesses whether or not the farmer has access to crop diversification and other related information. Access to such information encourages farmers to venture into diverse crops for consumption and income purposes. Through the extension service, it is expected that prior information on available markets for the crop of interest can be provided. Hence the variable can influence the choice of market channel for mixed beans. Kihoro (2016) found that farmers who had access to extension services easily accessed market information; therefore, they could sell to beneficial channels.

Access to loans: Is a dummy variable representing the farmer's access to credit facility for farming purposes from formal and informal sources. The variable takes the value of one if the farmer has access and zero if otherwise. Credit is known to boost agricultural production as it enables farmers to purchase the needed inputs, implements and labour. Access to credit is therefore hypothesised to influence channel choice of market outlet decisions. In a study by Urquieta (2009), access to loans was found to significantly influence the channel choice of market outlet for potatoes in Bolivia.

Cooperative membership: is a dummy variable with a value of one if the farmer is a member of a cooperative and zero if otherwise. Belonging to a cooperative is believed to help members source inputs and markets for the crop of interest. A cooperative is also a source of relevant production and market information that can help in the farmer's decision-making process. Furthermore, the cooperative is an avenue for bulking and selling to lucrative market channels, which is advantageous to members. Tumukunde (2018) posits that groups facilitate farmers to bulk their production and sell collectively to gain from economies of scale.

Location Factors

The Agro-Ecological Regions (AER) refer to the households's geographical locations of the households and are intended to capture advantages and disadvantages of the different locations with regard to channel choice of market outlet. The relationships revealed by the results are to be explained by the specific attributes of each of the locations. The differences in rainfall pattern in these areas determine the mixed beans production levels and channel choice of market outlets. Three dummy variables, one for Agro-Ecological Region I (AER I), one for Agro-EcologicalRegion II (AER II) and one for Agro-Ecological Region III (AER III), were defined. Agro-Ecological Region I (AER I) was coded as a dummy variable taking a value of one if the household is found in Agro-Ecological Region I and the value zero if otherwise. AER I is characterized by low rainfall (below 1000mm) and covers Zambia'sSouthern and Western parts. Because of the low rainfall found in this region, production of mixed beans is equally low. Therefore, households found in this region are likely to produce smaller quantities of beans, hence influencing channel outlet choice. Agro-EcologicalRegion III (AER III) was coded as a dummy variable, taking a value of one if the household is found in Agro-EcologicalRegion III and the value zero if otherwise. AER III is characterized by high rainfall per annum (1200mm and above) and lies in the country's Northern and North-Western parts, favouring beans production (Hamazakaza et al., 2016). Unlike farmers in AER I, farmers in this region are likely to produce higher quantities of mixed beans.

CHAPTER 3 - METHODOLOGY

3.0 Introduction

This chapter presents the methods and procedures used to achieve the objectives of the study. The section presents the source and type of data used. Also, the section presents econometric models used to analyse the factors influencing participation, the extent of participation and factors influencing the choice of marketing outlet of mixed beans in Zambia.

3.1 Data Type, Source and Data Collection Methods

This study used secondary data from the Rural Livelihoods Survey (RALS) of 2015. The RALS 2015 survey was part of a longitudinal study that had two waves at the time of this research i.e. RALS 2012 and RALS 2015. The study only used the RALS 2015 survey data hence being cross-section in design. The RALS 2015 continues from the RALS 2012 survey, which was based on the 2010 Census of Housing and Population sampling frame. Households were selected using a stratified two-stage sample design. The sampling covered all the ten provinces, and a total of 8,840 households were selected for the RALS 2012. With the RALS 2015, a total of 7,934 households were included in the sample. The two surveys were conducted by the Indaba Agricultural Policy Research Institute (IAPRI) in collaboration with ZAMSTATS, formerly the Central Statistics Office (CSO) and the Ministry of Agriculture (MA). The RALS data set was used because it contains a country-wide and broader set of detailed information on smallholder farmers' livelihood activities and outcomes. The sampling procedure involved selecting households growing mixed beans. Using this criterion, a sample of 1,326 smallholder farmers was included in the analysis.

3.2 Sample selection

The main objective of the study was to determine factors influencing participation in the mixed beans market among the bean growers in Zambia. Furthermore, the study aimed to determine factors that influence channel choice of market for the mixed beans growers. For this reason, only beans growing households from the RALS 2015 survey were selected giving a total sample of 1,326 households out of a total of 7,934 households.

3.3 Data Analysis

The study used STATA 15, SPSS 22 and MS Excel software packages for analyzing the data. The STATA 15 package was used for econometric analysis whilst the SPSS 22 package was used for data processing and descriptives. The MS Excel package was used for the generation of tables and graphs. The Heckman and MNL models were used in the econometric analysis, which are further described below.

3.4 Analytical Framework

The models used in this study are the Heckman model and the MNL. The Heckman model was used to evaluate the factors that influence the decision and extent of participation by smallholder farmers in the market of mixed beans. The Multinomial Logit Model was used to assess the factors that influence choice of marketing channel by smallholder mixed bean producers.

3.4.1 Factors Influencing Market Participation

The Heckman model was used to determine factors influencing participation and the extent of participation in the mixed beans market. The Heckman two-step procedure was considered as an appropriate model because of the anticipated problem of selection bias in the sample. Selection bias arose because not all households in the survey produced beans, i.e. 1,326 households out of 7,936 households in the survey were mixed beans producers. Furthermore, 67% of mixed beans producers participated in selling of beans; hence the anticipated selection bias in the data (RALS 2015). The Heckman is popular because it is a relatively simple procedure for correcting sample selection bias (Hoffman and Kassouf, 2005). The Heckman two-step procedure is an estimate that is based on two equations. In this context, the first equation involved estimating whether or not the farmer participates in the mixed beans market.

The first equation of the Heckman model was estimated using the Probit model, whilst the second equation (outcome equation) was estimated using OLS regression. The Probit model predicts the probability of whether the individual farmer participated in the mixed beans market or not as shown below:

$$pr\left(Z_{i}=1\left|w_{i,\alpha}\right)=\phi\left(\lambda\left(w_{i,\alpha}\right)\right)+u_{i}$$
(1)

Where Z_i is an indicator variable equal to unity for the household that participated in marketing, ϕ is the standard normal cumulative distribution function, w_i is the vector of factors affecting market participation and, α is the vector of coefficients to be estimated, u_i is the error term assumed to be distributed normally with a mean of zero and a variance σ^2 . The variable Z_i takes the value of 1 if the marginal utility the farmer *i* gets from participating in mixed beans marketing is greater than zero, and zero if otherwise. This is shown below;

$$Z_i^* = \alpha w_i + v_i \tag{2}$$

Where,

$$Z_i = 1 \text{ if } Z_i^* > 0$$

$$Z_i = 1 \text{ if } Z_i^* \le 0$$
(3)

The second step uses a regression model as shown below;

$$Y_i = X_i \beta + \varepsilon_i \tag{4}$$

Where X_i represents a vector of explanatory variables determining market intensity outlined in table 2, β is a vector of coefficients and ε_i the error term.

Table 1 below shows a summary of the explanatory variables hypothesised to influence market participation and extent of participation.

Dependent Variables

Type of household (Typhh): Is a dummy variable used in the Heckman model equation taking one if household is a market participant. One represents actual participation in the market by selling of mixed beans and zero if household produces but does not sell.

Quantity of mixed beans sold (kg_sold): Is a continuous dependent variable used in the outcome equation of the Heckman model measured in kilograms. It measures the actual quantities of mixed beans sold by market participants.

| Variable | Measurement of the Variables | Hypothesised Sign |
|--------------------------------------|---|-------------------|
| Dependent Variables | | |
| Type of household | Dummy (1 = market participant, 0 = Non-Market participant) | Dependent |
| Quantity of mixed beans sold | Kilograms (continuous) | Dependent |
| Independent Variables | | |
| Farmer and Household Characteristics | 1 | |
| Age in years | Years (continuous) | +/- |
| Maximum level of education | Years (continuous) | + |
| Gender | Dummy (1=male, 0=female) | + |
| Adult equivalent | Number (continuous) | +/- |
| Household Assets | | |
| Total land holding size | Hectare (continuous) | + |
| Production Factors | | |
| Total area cultivated | Hectare (continuous) | + |
| Area planted to beans | Hectare (continuous) | + |
| Quantity harvested | Kilograms (continuous) | + |
| Simpson Index of Diversity-Crops | Unit (continuous) | + |
| Tropical Livestock Unit (TLU) | Unit (continuous) | +/- |
| Net-off-farm income | Zambian Kwacha (Continuous) | +/- |
| Institutional Factors | | |
| Access to loans | Dummy $(1 = yes, 0 = otherwise)$ | + |
| Dummy access to price information | Dummy $(1 = yes, 0 = otherwise)$ | + |
| Dummy Cooperative membership | Dummy $(1 = yes, 0 = otherwise)$ | + |
| Dummy access to extension service | Dummy $(1 = yes, 0 = otherwise)$ | + |
| Distance to market | Kilometer (continuous) | - |
| Location Factors | | |
| Agro-Ecological_Region I (AER I) | Dummy (1=yes, 0=otherwise) | - |
| Agro-Ecological_Region III (AER III) | Dummy (1=yes, 0=otherwise) | + |
| Source: Adapted from literature | | |

Table 2: Summary of explanatory variables used in the Heckman two-stage model

To correct for potential selection bias, an additional regressor in the equation is included. The regressor is the Inverse Mills Ratio (IMR). The IMR is therefore computed as follows;

$$\frac{\varphi\left(h\left(w_{i},\tilde{a}\right)\right)}{\varphi\left(w_{i},\tilde{a}\right)} \tag{5}$$

Where φ is the normal probability density function. Hence, the second-stage equation is given by:

$$E = (Y_i | Z = 1) = f\left(x_{i,\beta}\right) + \lambda \frac{\varphi\left(\lambda\left(w_{i,\hat{a}}\right)\right)}{\varphi\left(w_{i,\hat{a}}\right)}$$
(6)

Where;

E is the expectation operator, *Y* is the (continuous) proportion of mixed beans sold by the farmer,

x is the vector of independent variables affecting the quantity of mixed beans sold, β is the vector of the corresponding coefficients to be estimated.

Hence, Y_i can be expressed as follows:

$$Y_i^* = \beta' x_i + y \lambda_i + u_i \tag{7}$$

 Y_i^* is only observed for those mixed beans farmers participating in the market. Where;

$$u_i \sim N(0, \sigma_u). (Z_i = 1)$$
, in which case $Y_i = Y_i^*$ (8)

The first step that involves the decision of whether or not to participate in mixed beans marketing, can hence be presented as follows:

$$P_{(0,1)} = \beta_0 X_0 + \beta_1 X_1 + \beta_2 X_2 + \dots + \beta_n X_n + e$$
(9)

Where participation is denoted by 1 and non-participation by 0. β_0 is a constant whilst β_1n are parameters to be estimated, X_s are vectors of explanatory variables and e is the error term.

Before running the Heckman two-step model, statistical tests were conducted on the data, which include the Variance Inflation Factor. Gujarat (2004) indicated that Variance

Inflation Factor (VIF) is used to check multicollinearity among continuous variables. If the value of VIF is greater than 10, the variables are said to be highly collinear.

VIF
$$(X_j)=1/(1-R_j^2)$$

(10)

Where R_j^2 is the multiple correlation coefficients between explanatory variables , the larger the value of R_j^2 , the higher the value of VIF(Xj) causing collinearity in the variable (Xj).

3.4.1.1 Model Specification

Heckman two-stage procedure

Heckman (1979) proposed a two-stage procedure which only involves the estimation of a standard Probit and a linear regression model. The exclusion restrictions were invoked in estimating the Heckman two-stage procedure. According to Certo *et al.*, (2016), Heckman models require exclusion restrictions in the first stage that do not appear in the second stage. Certo *et al.*, (2016), further contend that when the exclusion restrictions are poor in a model (or do not exist at all), IMR will correlate too highly with *x*, thus introducing multicollinearity problems in the second stage of the model (Sartori 2003). Before running the Heckman model, some diagnostic tests were carried out to check for multicollinearity.

The variance inflation factor (VIF) was used to test for multicollinearity and the results are shown in Appendix1.

The two equations for the two-stage procedure are therefore specified as follows:

Selection equation

Type of Household (market participant/non market participant) $(P_i(0, 1)) = \beta_0 + \beta_1 X_1 + \beta_2 X_2 + \dots + \beta_n X_n + e$ (11)

$$\begin{split} P_{i}(0,1) &= \beta_{0} + \beta_{1headage} + \beta_{2dhhsex} + \beta_{3ae} + \beta_{4edu_max} + \beta_{5landholdingsz15} + \\ \beta_{6mixedbeansha} + \beta_{7totharv} + \beta_{8SID_crops} + \beta_{9dinf_price} + \beta_{10net_{off_}farm} + \\ \beta_{11dist_market} + \beta_{12dadv_ext} + \beta_{13AERI} + \beta_{14AERIII} + \beta_{15dloan} + \beta_{16dcoop_mem} + \end{split}$$

Formatted: Heading 4, Left, Space Before: 0 pt, After: 0 pt, Line spacing: single

$\beta_{17TLU} + \varepsilon_i$

Outcome equation

Quantity of bean sold $(Y_i) = \beta_0 + \beta_1 X_1 + \beta_2 X_2 + \dots + \beta_n X_n + e$ (13)

Following the exclusion restrictions, the following variables were not included in the second stage of the Heckman Two-Stage Model. These include access to loan, membership to cooperative and Tropical Livestock Unit

The proportion of mixed beans sales:

$$Y_{i} = \beta_{0} + \beta_{1headage} + \beta_{2dhhsex} + \beta_{3ae} + \beta_{4edu_max} + \beta_{5landholdingsz15} + \beta_{6mixedbeansha} + \beta_{7totharv} + \beta_{8SID_crops} + \beta_{9dinf_price} + \beta_{10net_{off_farm}} + \beta_{11dist_market} + \beta_{12dadv_ext} + \beta_{13AERI} + \beta_{14AERIII} + \varepsilon_{i}$$
(14)

3.4.2 Factors influencing channel choice of market outlet

The Multinomial Logit model was used to determine factors influencing channel choice of market for mixed beans among Zambian smallholder farmers. The choice of a particular market outlet is discrete because it is selected among other alternative outlets.

Let P_{ij} represent the probability of choice of any given market outlet by mixed beans farmers. The equation will therefore be as follows;

$$P_{ij} = \beta_0 + \beta_1 X_1 + \dots \beta_k X_k + e$$
(15)

In this study, the farmers have four market outlets to sell their mixed beans produce, therefore *i* takes values (1, 2, 3, 4), each representing the choice of the market outlet (small scale traders =1, retailer/marketeers =2, other households =3, other buyers =4). The MNL model predicts the relative probability that a producer would choose one of the four categories based on the nature of the explanatory variables. X_i are factors affecting choice of a market outlet, β are parameters to be estimated and *e* is a randomized error. *j* are the alternative choices, and the probability of choosing outlet *j* is given by:

(12)

$$Prob(Y_i = j) = \frac{e_{z_j}}{\sum_{k=0}^{j} e_{zk}}$$
(16)

Where

 Z_i a choice and Z_k is an alternative choice that could be choosen (Greene 2000). The model estimates are used to determine the probability of choice of a market outlet given *j* factors that affect the choice X_i . With a number of choices log odds ratio is computed as follows;

$$\ln\left(\frac{P_{ij}}{P_{i^*}}\right) = \alpha + \beta_1 X_1 + \beta_2 X_2 + \dots + \beta_k X_k + e_i$$
(17)

Where:

 P_{ij} and P_{ik} are probabilities that a farmer will choose a given outlet and an alternative outlet respectively.

In $\left(\frac{P_{ij}}{P_{i^*}}\right)$ is a natural log of probability of choice *j* relative to probability choice *k*, where; α is a constant, β is a matrix of parameters that reflect the impact of changes in *X* on probability of choosing a given outlet, *e* is the error term that is independent and normally

distributed with a mean zero. The parameter estimates of the MNL provide only the direction of the effect of the

independent variable on the dependent variable but do not represent either the actual magnitude of change nor probabilities. The marginal effects or marginal probabilities are functions of the probability itself and measure the expected change in the probability of a particular choice being made with respect to a unit change in an independent variable from the mean (Green 2000).

The marginal effects of the attributes on choice are determined by getting the differential of probability of a choice and it is given by;

$$(\delta) = \frac{dP_i}{dX_i} = pi(\beta_j - \sum_{k=0}^j P_k \beta_k = P_i(\beta_i - \beta)$$
(18)

Therefore, the MNL Model is given as follows:

Market Channel Outlet (P_{ij}) = $\beta_0 + \beta_1 X_1 + \beta_2 X_2 \dots \beta_n X_n + \varepsilon_i$ (19)

Choice of market outlet:

The Multinomial logit model was specified as a function of 17 independent variables influencing mixed beans farmers' market channel choices:

 $P_{ij} = \beta_0 + \beta_{1headage} + \beta_{2dhhsex} + \beta_{3ae} + \beta_{4edu_max} + \beta_{5landholdingsz15} + \beta_{6mixedbeansha} + \beta_{7totharv} + \beta_{8SID_crops} + \beta_{9dinf_price} + \beta_{10net_{off_f}arm} + \beta_{11dist_market} + \beta_{12dadv_ext} + \beta_{13AERI} + \beta_{14AERIII} + \beta_{15dloan} + \beta_{16dcoop_mem} + \beta_{17TLU} + \varepsilon_i$ (20)

The explanatory variables included in the above model were: age, gender, and education level of the household head, adult equivalent, landholding size, area planted, quantity harvested, Simpson Index of Diversity (SID-Crops), distance to nearest market, TLU, access to price information, net off-farm income, cooperative membership, access to loans, access to extension service, Agro-ecological region I, and Agro-ecological region III.

Table 2 below shows a summary of the explanatory variables which are hypothesised to influence the dependent variable.

The Dependent Variable

Marketing channel outlet (Channel): Is a variable that is measured by the probability of selling mixed beans to either of the markets. The dependent variable is a discrete choice in the MNL model, with the farmers grouped into four categories based on their pattern of choice of market for their produce. The farmers' choice involves selecting from among the four channels which are the model's dependent variables for the model. In the model, Y_1 represents farmers who choose to sell to "small scale traders" (base category), Y_2 represents farmers who choose to sell to "retailer/marketeers", Y_3 represents farmers who choose to sell to "Other households", and Y_4 represents farmers who choose to sell to "Other buyers".

Table 3: Summary of Variables used in the MNL Model

| Variable | Measurement of the Variables | Hypothesised Sign | |
|---|---|-------------------|--|
| Dependent Variable: | | | |
| Households' channel choice of mixed beans market outlet | (Small scale traders = 1 [base category], Retailer/marketeer = 2, Other households = 3, Other buyers = 4) | Dependent | |
| Independent Variables: | | | |
| Farmer and Household Characteristic | s | | |
| Age in years | Years (continuous) | +/- | |
| Maximum level of education | Years (continuous) | + | |
| Gender | Dummy (1=male, 0=female) | + | |
| Adult equivalent units | Number (continuous) | + | |
| Household Assets | | | |
| Total land holding size | Hectare (continuous) | + | |
| Production Factors | | | |
| Total area cultivated | Hectare (continuous) | + | |
| Area planted to beans | Hectare (continuous) | + | |
| Quantity harvested | Kilograms (continuous) | + | |
| Simpson Index of Diversity-Crops | Unit (continuous) | + | |
| Value of production assets | Zambian Kwacha (continuous) | + | |
| Tropical Livestock Unit (TLU) | Unit (continuous) | +/- | |
| Off-farm income | Dummy $(1 = yes, 0 = otherwise)$ | +/- | |
| Institutional Factors | | | |
| Access to loans | Dummy $(1 = yes, 0 = otherwise)$ | + | |
| Dummy access to price information | Dummy $(1 = yes, 0 = otherwise)$ | + | |
| Dummy Cooperative membership | Dummy $(1 = yes, 0 = otherwise)$ | + | |
| Dummy access to extension service | Dummy $(1 = yes, 0 = otherwise)$ | + | |
| Distance to market | Kilometer (continuous) | - | |
| Location Factors | | | |
| Agro-Ecological_Region I (AER I) | Dummy (1=yes, 0=otherwise) | - | |
| Agro-Ecological_Region III (AER III) | Dummy (1=yes, 0=otherwise) | + | |

Source: Adapted from literature

CHAPTER 4 – RESULTS AND DISCUSSION

4.0 Introduction

This chapter initially presents the characteristics of smallholder bean growers in Zambia. The smallholder bean growers are categorized into mixed beans market participants and non-market participants and overall mixed beans producers. After that, results and discussion from econometric analysis on factors that influence market participation and extent of participation, and market channel choice for beans are presented.

4.1. Characteristics of Sample Households

The characteristics of the farm household head and the household at large are likely to influence the decision to participate or not to participate in the beans market. To better understand factors that influence market participation, the bean producers were classified into market participants and non-market participants. The market participants refer to bean producers that sold their beans whilst the non-participants were the category that did not sell. For continuous variables, the t test was used to determine whether there was any significant difference between the two groups of market participants and non-participants. The chi square test was also used to determine the significant difference for categorical variables between the two groups. The characteristics of the 1,326 mixed beans producers are presented in Table 4 and Table 5.

Mean age results indicate that there was no significant difference between the market participants and non-market participants. Market participants had an average of 48.19 years, and 49.52 years for non-market participants. Adult equivalent results show that there was no statistical difference between the two groups. The Adult equivalent result for the market participants was 5.29 and 5.31 for the non-market participants.

Education level results did not show any significant difference between the two groups. On average, market participants had achieved 8.68 years of schooling, whilst the nonmarket participants had achieved 8.67 years. The implication is that most bean producers had reached secondary education level. Makhura *et al.*, (2001) state that human capital represented by the household head's formal education is expected to increase a household's understanding of market dynamics and improve decision about the amount of output sold. Table 3 below shows bean producers' characteristics in terms of continuous variables

| Variable Name (Continuous variables) | Total Sample (N=1,326) | Market Participant (N=894) | Non-Market Participant (N=432) | T-Test | | | |
|---|--------------------------------------|----------------------------------|--------------------------------------|-----------|--|--|--|
| | Mean | Mean | Mean | | | | |
| Farmer and Household Characteristic | Farmer and Household Characteristics | | | | | | |
| Age (Yrs) | 48.62 | 48.19 | 49.52 | 1.62 | | | |
| Adult Equivalent Units (Units) | 5.31 | 5.29 | 5.31 | -0.13 | | | |
| Education level (Yrs) | 8.67 | 8.68 | 8.67 | -0.1 | | | |
| Land holding (Ha) | 6.92 | 7.67 | 5.35 | -2.69** | | | |
| Land cultivated (Ha) | 3.03 | 3.2 | 2.68 | -3.61** | | | |
| Area planted (beans) (Ha) | 0.43 | 0.51 | 0.26 | -8.23*** | | | |
| Quantity harvested (Kg) | 199.45 | 257.74 | 77.13 | -8.87*** | | | |
| Volume sold (Kg) | 120.79 | 179.16 | 0 | -11.18*** | | | |
| Simpson Index of Diversity (SID) | 0.54 | 0.56 | 0.51 | -5.45*** | | | |
| Net off-farm income (ZMW) | 7,779.87 | 7,094.47 | 9,198.25 | 1.55 | | | |
| Livestock holding (TLU) | 2.9 | 2.55 | 3.62 | 1.79* | | | |
| Production assets (ZMW) | 22,342.09 | 20,430.75 | 26,353.20 | 1.22 | | | |
| Market Access & Institutional factors | | | | | | | |
| Distance to market (Km) | 25.72 | 26.24 | 24.62 | -0.84 | | | |

<u>Table 4</u>: Household Characteristics of Bean Producers (Continuous Variables)

Note: ***Significant at 1 per cent; **Significant at the 5 per cent level *Significant at the 10 per cent level

Source: Owner's own computation

Landholding size results show a significant difference between market participants and non-market participants. Market participants had an average of 7.67 hectares, whilst non-participants had an average of 5.35 hectares. Also, land under cultivation results show a significant difference between the two groups. Market participants cultivated an average of 3.2 hectares, whilst non-market participants only had 2.68 hectares. Cultivating more land entails that the farmer is likely to increase the area allocated to the crop of interest. In addition, increasing land cultivated increases the probability of larger harvests and subsequent surplus for sell.

Land planted to beans results show a significant difference between market participants and non-market participants. Market had a higher proportion of land planted to beans averaging 0.51 hectares, whilst non-market participants only had 0.26 hectares. Allocating larger area to beans leads to a subsequent increase in the harvest.

Quantity harvested results indicate a significant difference between market participants and non-market participants. Market participants had higher average quantities harvested at 257.74 Kg, non-market participants had 77.13 Kg. This means that larger harvests lead to increased chances of participating in the mixed beans market. Mean volumes sold also show a significant difference between the two groups. The market participants sold an average 179.2 kilograms, whilst non-market participants did not sell (0 kilograms). This shows that only participating farmers could sell beans compared to the non-participating farmers who did not sell.

The Simpson Index of Diversity (SID) results show a significant difference between the two groups. Market participants show a higher score of 0.56, whilst the non-market participants show a score of 0.51. This shows that market participants practiced crop diversification than the non-market participants. The SID is used to measure the degree of crop diversity, and a score closer to 1 indicates that there is increased diversity in terms of the number of crops grown (Rahman *et al.*, 2009).

In addition to farming activities, the smallholder mixed bean farmers engage in off-farm activities, which contribute to their overall financial position. Net off-farm income results did not show any significant difference between the two groups. Market participants earned an average of ZMW 7,094.47, whilst non-market participants earned an average ZMW 9,198.25. Some studies (Mirie and Zemedu, 2018) suggest that non-farm income has a positive influence on the decision to participate in a market.

Livestock holding (TLU) results show that there was a significant difference between the two groups. Market participants had a lower TLU index of 2.55, whilst non-market participants had an index of 3.62. This indicates that non-market participants had more livestock than the market participants. Value of productive assets results did not show any statistical difference between the two groups. Productive assets include land, farm equipment and livestock (Kumaraswamy *et al.*, 2020.), which enhance productive capacity of farmers leading to larger harvests and access to transport.

Distance to market for bean results show that there is no significant difference between market participants (26.2 kilometres) and non-market participants (24.6 kilometres) Distance is one of the crucial factors that determine market participation. According to Tesfew (2013), farmers whose residences are closer to the market would have more access and a higher level of market participation than those who live away from the nearest market centre. This is because distance coupled with poor roads characteristise of rural areas can be associated with high transport costs.

The t test results revealed that there were significant differences between the mixed beans market participants and non-market participants for seven variables of landholding size, land cultivated, area planted, quantity harvested, quantity sold, SID-Crops, and TLU. However, t test results for six variables for age, adult equivalent, education level, net-off farm income, production assets and distance to market did not portray any significant difference between the two groups. Table 4 below shows characteristics of bean producers in terms of categorical variables

| Variable Name (Categorical Variables) | Total Sample (N=1,326) | Market Participant (N=894) | Non-Market Participant (N=432) | Chi Square | | |
|--|------------------------------|----------------------------------|--------------------------------------|---------------|--|--|
| | % | % | % | | | |
| Farmer and Household Characteristics | 1 | | | | | |
| Gender (=1 if male) | 0.83 | 0.85 | 0.81 | 3.36* | | |
| Decision maker (=1 if female) | 0.24 | 0.15 | 0.36 | 12.66*** | | |
| Market Access & Institutional factors | | | | | | |
| Access to price information (=1 if yes) | 0.75 | 0.76 | 0.71 | 5.15** | | |
| Access to extension service (=1 if yes) | 0.58 | 0.58 | 0.58 | 0.01 | | |
| Cooperative member (=1 if yes) | 0.68 | 0.7 | 0.63 | 5.60** | | |
| Access to loan (=1 if yes) | 0.09 | 0.08 | 0.1 | 0.89 | | |
| Location factors | | | | | | |
| AER I | 0.04 | 0.01 | 0.08 | 35.072*** | | |
| AER II | 0.15 | 0.12 | 0.408 | 18.97*** | | |
| AER III | 0.82 | 0.87 | 0.71 | 46.53*** | | |
| Note: ***Significant at 1 per cent: **Significant at the 5 per cent level *Significant at the 10 per | | | | | | |

Table 5: Household Characteristics of Bean Producers (Categorical Variables)

Source: Owner's own computation

cent level

Gender results show that 85% of market participants are male-headed, whilst 81% of nonmarket participants are male-headed. Traditionally, most households in Zambia are maleheaded. Furthermore, results for market participants show that only 15% of decisionmakers on whether to produce bean or not in a household are female. However, nonparticipants results show a higher percentage of 36% of females being decision-makers on whether to produce bean or not in the household.

Access to price information results indicate a significant difference between market participants (76%) and non-market participants (71%). Prior information on the price of a commodity influences the production decision. Furthermore, price information assists the bean producer make informed decisions on selling. Access to extension service results show no significant difference between market participants and non-market participants. Access to extension service assists farmers increase their knowledge in production and marketing.

Membership to cooperatives show significant difference between market participants (70%) and non-market participants (63%). Cooperatives are sources of inputs and relevant information that contribute to increased production. Access to loan results show that there was no significant difference between market participants (8%) and non-market participants (10%). Loan access improves the farmers' ability to acquire, among others, inputs and labour to improve production and productivity. The result show that few mixed beans farmers access loans to support their agricultural production.

Geographical location is one of the critical factors in the production of beans in Zambia. AER I results show that there was significant difference between the two groups. Market participants in AER I were only 1%, whilst non-market participants were 8%. AER II results indicate a statistical difference between the two groups. Market participants in AER II were only 12%, whilst non-market participants were 40.8%. AER III results also indicate a significant difference between the two groups. Market participants in AER III were 87%, whilst non-market participants were 71%. It is therefore expected that mixed bean producers in AER III are likely to participate in the market.

The chi square test portrayed significant results for ten categorical variables apart from the three categories of primary education, access to extension services, and access to loan. A scatterplot was used to assess the bivariate relationship between the quantity harvested and quantity sold for mixed beans. Figure 3 shows the possible relationship between two variables.



Figure 2: Scatterplot of quantity harvested and quantity sold

Source: Authors own computation

Results from Figure 2 show a positive relationship between quantity harvested and quantity sold for mixed beans. The data shows an uphill pattern as we move from left to right. In addition, the pattern assumes a linear form; hence, it can be hypothesised that as quantity harvested increases, quantity sold also increases. This entails that farmers with larger bean harvests are more likely to sell than those with relatively lower harvests.

4.1.2 Characterizing mixed beans households by market channel

In the current study, mixed bean farmers are faced with four choices of market channels, these are small scale traders, retailer/marketeer, other households, and other buyers. The small scale trader was taken as the base category to analyze the determinants of the farmer's choice of market channel. In this study, the other buyers outlet comprise large scale traders, NGOs/Faith Based Organizations/Churches, Cooperatives, Community Markets for Conservation (COMACO), Government institutions (schools, hospitals/health centres). Other buyers are characterized by purchasing mixed bean in bulk from the farmer and entering into contractual arrangements.

Figure 3 shows the preference of market channels for beans. Results indicate that "Smallscale traders" are the most preferred market channel with 474 households (53%). This could be attributed to the high mobility of small-scale traders. Small-scale traders can move from place to place in search of merchandise of interest. In this way, the farmers' transaction costs are minimized as they do not incur transportation costs for the produce. In addition, the frequent interaction between the small-scale traders and mixed beans growers contributes to building long standing business relationships. Anh and Bokelmann (2019) note that repeated economic transaction embedded in the social relationship indicates the largest sales volume".

The "Other households" market channel was the second most preferred market outlet with 205 households (23%). This could be that other households are usually found near the beans farmers. The close proximity helps to reduce transaction costs for the beans farmers. The "Retailer/Marketeer" is the third most preferred market channel with 175 households (20%). The retailers/marketeers are usually located at market places or common trading places and buy moderate quantities for resale in smaller quantities directly to consumers. Farmers have to transport the beans for sale to the retailer/marketeer hence incurring transaction costs.



Figure 3: Most preferred market channel for mixed beans

Source: Authors own computation

The least preferred market channel is the "Other buyers" with 38 households (4%). This channel is composed of large scale buyers, institutions (schools, health institutions), NGOs and private companies. The "Other buyers" are known to purchase large quantities from individual farmers. In addition, the other buyers are known to enter into contracts or legal agreements with farmers to assure ready market and other obligations between parties. It can, therefore, be assumed that only farmers who produce large quantities and understand contractual obligations manage to trade with "other buyers" market channel.

4.2 Econometric Model Outputs

4.2.1 Factors influencing market participation

The Heckman two-step procedure was used to analyse the determinants of market participation among smallholder mixed bean producers. Before running the Heckman model, some diagnostic tests such as the VIF were carried out to check for multicollinearity in the data. The results indicate a VIF less than 10. The variables included in the model were household head's age, education level, adult equivalent, TLU,

landholding size, area planted to beans, quantity of beans harvested, SID, access to price information, access to extension services, distance to market for beans, access to loans, cooperative membership, AER I, and AER III. Post estimation of the selection equation was carried out after analysing the data, this was done to obtain marginal effects, which were used for interpretation.

The results indicate that six variables (education level, quantity harvested, access to price information, AER I, AER III, and net off-farm income) significantly influenced the smallholder farmers' decision to participate in the mixed bean market. The Inverse Mills Ratio (IML/Lambda) term was significant and positive at (0.004) suggesting that the error term in the selection and outcome equations is positively correlated. The implication is that unobserved factors making participation in mixed bean marketing are more likely to be associated with higher scores on the dependent variable. Table 5 below presents results of the selection equation, marginal effects.

Age of the household head negatively and significantly influenced market participation. An increase in the age of household head by one year decreases the probability of participating in the mixed bean market by 0.2%, all other factors held constant. The implication is that younger people are more enthusiastic about participating in the mixed bean market than older people. This is in line with Sigei (2014) and Sebatta *et al.*, (2013) findings, that age had a significant but negative influence on the decision to participate in the pineapple and potato markets in Kenya and Uganda respectively.

Results revealed that education level positively and significantly influences the decision to participate in the mixed beans market. An increase in one year of schooling increases the probability to participate in the mixed beans market by 0.8%, all other factors held constant. This could be that formal education increases the managerial aspects, leading to increased production and marketing skills, which enable the farmer to participate in the market. The results are consistent with Mango *et al.*, (2018) who found a positive relationship between level of education and participation in the rice market.

<u>Table 6</u>: Factors affecting decision to participate in the mixed bean market (Heckman Selection equation results)

| Variable | Marginal effects | Std. Err. | P>z |
|------------------------------|------------------|-----------|-------|
| Age | -0.001 | 0.001 | 0.172 |
| Gender | 0.013 | 0.033 | 0.703 |
| Adult Equivalents | 0.000 | 0.006 | 0.938 |
| Education level | 0.007* | 0.004 | 0.099 |
| Landholding soze | 0.002 | 0.001 | 0.258 |
| Area planted | 0.008 | 0.044 | 0.856 |
| Quantity harvested | 0.001*** | 0.000 | 0.000 |
| Access to loans | -0.015 | 0.043 | 0.719 |
| Cooperative membership | 0.005 | 0.026 | 0.844 |
| SID _crops | 0.122 | 0.075 | 0.102 |
| Access to price information | 0.047* | 0.028 | 0.089 |
| Distance to market | 0.000 | 0.000 | 0.490 |
| Access to extension services | 0.023 | 0.025 | 0.371 |
| TLU | -0.001 | 0.002 | 0.521 |
| AERI | -0.210*** | 0.069 | 0.002 |
| AERIII | 0.139*** | 0.034 | 0.000 |
| Net off-farm income | 0.000*** | 0.000 | 0.008 |
| /Mills | | | |
| Lambda | 36.195 | 12.514 | 0.004 |
| rho | 0.514 | | |
| sigma | 70.394 | | |

***Significant at 1 per cent; **Significant at the 5 per cent level *Significant at the 10 per cent level

Source: Authors own computation

Results further show that the quantity harvested positively and significantly influenced the decision to participate in the mixed bean market. As expected, an increase in quantity harvested by one-kilogram increases the probability of participating in the mixed bean market by 0.2%, all other factors held constant. This implies that as mixed bean yield increases, participation in the market also increases. The results are consistent with Mbitsemunda and Karangwa (2017), who found that the quantity produced positively influenced the household's decision to participate in the bean market.

Agro-ecological region I (AER I) significantly but negatively influence the decision to participate in the mixed bean market. AER I's negative coefficient indicates that relative to AER II, farmers in AER I are less likely to participate in the mixed bean markets.

As expected, Agro-ecological region III (AER III) positively and significantly increased the decision of mixed bean producers to participate in the bean market. The positive coefficient on AER III indicates that relative to AER II, farmers in AER III are more likely to participate in the mixed bean market.

4.2.2 Factors influencing the extent of participation in the mixed bean market

The OLS regression was estimated in the outcome equation. The dependent variable was the extent of market participation measured by the quantity of output (kg of beans) sold. The Wald Chi goodness of fit measured by Prob>Chi2 =0.000 showed that the explanatory variables used in the second stage of the Heckman procedure explained the variations in the extent of participation.

The results of the analysis are presented in Table 6 below and explanation on the factors that were statistically significant follow. The positive and negative significant relationships in the result indicate that a unit increase or decrease in the variables will lead to an addition or reduction in the quantity of mixed beans sold. Seven variables (household head's age, adult equivalent, household head's education level, quantity of beans harvested, TLU and Agro-Ecological Region I (AER I) significantly influenced the volume of mixed beans sold to the mixed bean market.

The coefficient of age significantly but negatively influenced the extent of market participation. An increase in the household head's age by one year reduces the quantity of mixed bean supplied to the market by 0.55 Kg; all factors held constant. The possible explanation could be that younger farmers are more energetic and enthusiastic and, therefore, can participate in beans marketing than the older farmers. Besides, older farmers tend to have more children and dependents, which increases consumption thereby reducing the quantities for sale.

Table 7: Factors influencing the extent of participation in the mixed bean market

(Heckman outcome equation results)

| Variable | Coef | Std. Err. | P>z | |
|-----------------------------|---------------------|-----------|-------|--|
| Age | -0.550*** | 0.187 | 0.003 | |
| Gender | 0.580 | 6.951 | 0.933 | |
| Adult equivalent | -4.870*** | 1.242 | 0.000 | |
| Education level | 2.476*** | 0.898 | 0.006 | |
| Landholding size | 0.187 | 0.165 | 0.258 | |
| Area planted | 18.792*** | 4.951 | 0.000 | |
| Quantity harvested | 0.789*** | 0.007 | 0.000 | |
| Access to loans | 1.697 | 8.713 | 0.846 | |
| Cooperative membership | -2.945 | 5.389 | 0.585 | |
| Simpson Index of Diversity | -6.173 | 16.346 | 0.706 | |
| Price information access | 2.042 | 5.782 | 0.724 | |
| Distance to nearest market | 0.080 | 0.072 | 0.266 | |
| Extension service access | 5.772 | 4.978 | 0.246 | |
| TLU | -1.127*** | 0.421 | 0.008 | |
| AER I | -37.783* | 20.559 | 0.066 | |
| AER III | -8.424 | 8.036 | 0.295 | |
| No. obs = 1283 | Selected $= 867$, | | | |
| Wald Chi2 $(16) = 14249.22$ | Prob > Chi2 = 0.000 | | | |

* significant at 10% level; ** significant at 5% level; *** significant at 10% level Source: Author's own computation

Adult equivalent unit negatively and significantly influenced the extent of market participation. An increase in adult equivalents by a unit reduces the quantity of mixed bean for sale by 4.87 Kg; all other factors held constant. This entails that family labour force has an inverse relationship with quantity for sale. This could be that a larger family labour force requires larger amounts for consumption, which in turn reduces the quantities available for sale. This confirms Beadgie & Reddy (2020)'s finding, that family size (measured in adult equivalent) negatively influenced the supply of maize to the market.

Education level significantly and positively influenced the extent of market participation. An increase in education by one year increases the quantity of mixed bean sales by 2.48 Kg; all factors held constant. Mango *et al.*, (2018) argue that a higher education level has positive implications for the ability to understand and interpret extension information. This leads to increased surplus production and subsequent proportion available for sale. Area planted positively and significantly influenced the extent of market participation. A unit increase in area planted by 1 hectare increases the quantity of mixed bean sales by 18.79 Kg; all factors held constant. Therefore, increasing area planted leads to increased quantities harvested and subsequent proportion of mixed beans for sale.

As expected, quantity harvested significantly and positively influenced the extent of market participation. A unit increase in quantity harvested by 1 kilogram increases the quantity of mixed bean sales by 0.79 Kg; all factors held constant. Therefore, increasing the quantity harvested increases the subsequent proportion of mixed beans that can be for sale. This result conforms to Apind (2015) result who found that an increase in level of rice output produced by the household implied an increase in the marketable surplus thus increasing the extent of market participation.

TLU negatively and significantly influenced the extent of market participation. An increase in livestock owned by a unit reduces the quantity of mixed bean sold by 1.13 Kg; all factors held constant. This shows an inverse relationship with bean production and marketable surplus. Rehima (2006) study found a negative relationship between livestock ownership and quantity of pepper sales in Ethiopia.

AER I negatively and significantly influenced the extent of market participation. The negative coefficient entails being in AER I reduces the quantity expected to be sold by 37.78 Kg; all factors held constant.

4.2.3 Factors affecting the choice of mixed bean marketing channels

In identifying factors influencing determinants of marketing channel of choice for beans, the study used the MNL model. Small-scale traders were taken as the base category because it had more observations. The MNL gives a McFadden's R^2 (Pseudo R^2) of 0.1154 which means that the explanatory variables explain 11.5% of the variation in the dependent variable. The estimated probability was greater than the chi-square value (P>Chi2 = 0.000), which implies that the variable were jointly significant in explaining the dependent variable (Table 7)

Table 8: Marginal effects of factors influencing choice of market channel outlet

| Variable | Retailer/Marketeer | | Other Households | | Other Buyers | |
|-----------------------------------|------------------------|-------|------------------|-------|---------------------|-------|
| | Marginal effects | P>z | Marginal effects | P>z | Marginal effects | P>z |
| Age | -0.0001 | 0.955 | 0.0009 | 0.164 | 0.0002 | 0.613 |
| Gender | 0.059 | 0.149 | -0.0202 | 0.491 | 0.0078 | 0.569 |
| Adult equivalent | -0.005 | 0.595 | -0.0013 | 0.794 | -0.0017 | 0.553 |
| Education level | 0.011* | 0.074 | -0.0031 | 0.400 | 0.005** | 0.013 |
| SID | 0.014 | 0.903 | -0.1392** | 0.030 | -0.0471 | 0.137 |
| Landholding size | 0.001 | 0.343 | 0.0004 | 0.502 | -0.0007 | 0.205 |
| Area planted | 0.014 | 0.632 | -0.1653*** | 0.001 | -0.0049 | 0.300 |
| Quantity harvested | 0.0002*** | 0.000 | -0.0006*** | 0.000 | 0.0000 | 0.823 |
| Price information access | 0.070** | 0.048 | -0.0329 | 0.222 | -0.0034*** | 0.002 |
| Loan access | -0.016 | 0.773 | 0.0056 | 0.882 | 0.0031 | 0.879 |
| Cooperative membership | -0.057 | 0.125 | 0.0177 | 0.391 | 0.0151 | 0.223 |
| Distance to market | -0.002** | 0.012 | -0.0001 | 0.841 | -0.0001 | 0.390 |
| Net off-farm income | 0.066 | 0.347 | 0.000*** | 0.004 | 0.0000*** | 0.005 |
| Extension service access | 0.066** | 0.040 | 0.0108 | 0.595 | 0.0108 | 0.362 |
| TLU | -0.008 | 0.218 | 0.0024 | 0.294 | 0.0002 | 0.796 |
| AER I | -0.149* | 0.080 | 0.5301*** | 0.003 | -0.0396*** | 0.000 |
| AER III | -0.040 | 0.452 | 0.0212 | 0.467 | -0.0001 | 0.994 |
| Number of obs. $= 867$ | Wald chi2(51) = 924.73 | | | | | |
| Prob>chi2 = 0.000 | Pseudo $R2 = 0.1154$ | | | | | |
| Log pseudolikelihood = -862.79885 | | | | | | |

Note: ***, ** and * represents significance at 1%, 5% and 10% probability levels respectively.

Source: Authors own computation

The household head's education level significantly affected the choice of retailer/marketeer and other buyers market outlets. An additional year in school was associated with a 1.1% and 0.5% increase in the probability of choosing retailer/marketeer and other buyers' market outlets, respectively, relative to small scale traders (base category). This could be because education is associated with efficiency and better understanding of marketing strategies leading to higher profits. The Retailer/Marketeer and other buyers outlets offered a higher unit price compared to the small scale trader. Mango *et al.*, (2018) note that education levels affect market information interpretation, and hence, marketing channel choice by farmers. A study by Anteneh *et al.*, (2011) found that the level of education of the household head significantly influenced the choice of the coffee market outlet.

Simpson index of diversity (SID) negatively influenced the choice of other households outlet relative to the small-scale trader. A unit increase in the index for crop diversity

reduced the probability of choosing the other household outlet by 13.9% relative to the small scale trader outlet. A farmer who cultivates diverse crops is expected to earmark the crop that fetches higher income for the market whilst reserving the other crops for consumption. In Zambia, beans are considered one of the most profitable crops (Sichilima *et al.*, 2016). Since the farmer has other crops for consumption, they can then sell more of the mixed bean to bulk buyers such as the small-scale trader to earn higher income.

As expected, area planted significantly affected the choice of other households outlet. The probability of choosing the other household outlet decreased by 16.5% for every hectare planted compared to small scale traders. This implies that as area planted increases, harvests will also increase hence, the farmer may opt for a channel which will purchase in bulk to avoid postharvest losses. The small scale traders purchase in bulk compared to other households.

As expected, quantity harvested had a significant influence on the choice of the retailer/marketeer, other households, and other buyers market outlets. An increase in quantity harvested by one kilogram increased the probability of choosing the retailer/marketeer and other buyers market outlets by 0.02% and 0.03% respectively, relative to the small scale trader outlet. On the other hand, an increase in quantity harvested by one kilogram reduced the probability of choosing the other households market outlet by 0.06%, relative to the small scale trader. Similar to area planted, the implication is that as farmers produce more, they will seek channels that offer better prices and also purchase in bulk. A study by Sigei (2014) found that quantity harvested increased the opportunity of farmers with larger produce to sell at the market places than those with little produce.

As expected, access to price information access had a positive influence on the choice of the retailer/marketeer and other buyers market outlets. Access to price information increased the farmer's probability of choosing the retailer/marketeer market by 7.0% relative to the small-scale trader outlet. However, access to price information reduced the farmer's probability of choosing the other buyers' outlet by 0.34% relative to the small-scale traders outlet. Prior access to price information enhances the farmer's ability to make informed decisions about the choice of a market outlet offering a better price. The retailer/marketeer market outlet offered a higher price per kilogram of beans compared to

the small scale trader. Tola (2013) notes that farmers' marketing decisions are based on market price information, and poorly integrated markets may convey inaccurate price information, leading to inefficient product movement. A study by Negeri (2015) found access to price information to be statistically significant in the choice of the cooperative market outlet for coffee in Ethiopia.

As expected, a one kilometer increase in distance to nearest market reduced the farmer's likelihood of selling to the retailer/marketeer outlet by 0.2% relative to small scale traders. This entails that distance is associated with the choice of a market channel because small scale traders buy at farm gate hence reduce costs of transporting to markets. The finding concurs with Chalwe (2011) who found that distance to market was a significant factor influencing channel choice of market outlet for beans.

Net off-farm income positively influenced the choice of other households and other buyers outlets relative to the small scale traders outlet. An increase in net off income by ZMW 1 increases the probability of selling to the other households and other buyers by 0.0002% and 0.0005% respectively. This could be that farmers in off-farm activities have less time for on-farm activities hence produce less, which can only cater for other households compared to larger buyers. Another possible justification is that farmers engaged in off-farm activities establish networks with bulk buyers over time, hence opt to sell to them. A study by Mamo and Dagnet (2012) also found that off-farm income had a statistically significant influence on the choice of the livestock market outlet in Ethiopia.

Extension service significantly affected the choice of retailer/marketeer outlet relative to the small scale trader outlet. An increase in access to extension services was associated with 6.6% increase in the probability of selling in the retailer marketeer outlet. This is because extension services increases farmers access to vital information on production and marketing on crops of interest. The marketing information enhances the farmer's ability to choose channels offering better prices. The retailer/marketeer outlet had a higher price per kilogram for bean compared to the other market outlets. A study by Kihoro (2016) found that farmers who had access to extension services easily accessed market information, therefore, they could sell to lucrative marketing outlets.

Agro-Ecological Region I (AER I) significantly influenced the farmer's choice of the retailer/marketeer, other households and other buyers outlet. Agro-Ecological Region I was associated with 14.9% and 3.9% reduced probability of choosing the retailer/marketeer and other buyers market outlets respectively compared to the small scale trader outlet. On the other hand, AER I increased the probability to sell to the other households market outlet by 53% relative to the small scale trader outlet. The possible justification is that farmers in AER I are likely to have small harvests, which influences their selling to other households. Other households purchase in small quantities compared to bulk buyers such as the small scale traders.

CHAPTER FIVE - CONCLUSIONS AND RECOMMENDATIONS

5.1 Introduction

This chapter presents conclusions on factors influencing market participation and channel choice decisions in the mixed bean market. Policy recommendations are made concerning the study findings to enhance market participation and choice of market outlets among bean farmers in Zambia.

5.2 Conclusions

Understanding market participation and the extent of participation, including market channels, are essential to help mixed beans farmers achieve commercialization and market integration. Several socio-economic and institutional factors predispose farmers' ability to participate in the market, the extent of participation, and choice of a particular market channel.

Econometric results of this study show that the main determinants of market participation were Socio economic factors (household head's education level, quantity harvested, net off-farm income), institutional factors (access to price information, access to extension services), and location factors (AER III). These variables were found to positively influence the decision to enter the mixed beans market. These results differ from Chalwe (2011) study who only found price and price barter to be the positive determinants in influencing the decision to participate in the bean market in Zambia. The current study further shows that the extent of participation in the mixed beans market was positively influenced by the household head's education level and as expected, area planted to beans and quantity harvested.

The results from the MNL model show that factors such as the household head's education level, area planted, quantity harvested, net off-farm income, access to price information, access to extension services, distance to the market for beans, and AER I were significant. The determinants of the choice of the retailer/marketeer market outlet were education level, where more years schooling increased the probability of choosing the retailer/marketeer, quantity harvested, where larger harvests increased the probability of choosing the retailer/marketeer channel, access to price information, where increased access to price information increased the probability of choosing the retailer/marketeer

channel, access to extension services, where increased access to crop diversification information increased the probability of choosing the retailer/marketeer channel. However, other factors such as distance to the market and AER I reduced the probability for farmers to sell to the retailer/marketeer outlet.

The main determinants for farmers' choice of other households outlet over small-scale traders were, net off-farm income, where higher net-off farm income earned increased the probability of choosing other households, and AER I, where AER I region increased the probability of choosing the other households outlet. Other factors such as the SID, and an increase in area planted and quantity harvested reduced the probability to choose the other household market outlet.

The study shows that the main determinants of farmers' choice of other buyers outlet over the small-scale traders were, education level, where more years on schooling increased the probability to choosing the other buyers outlet, quantity harvested, where larger harvests increased the probability to choose other buyers, net off-farm income, where earning larger incomes from off-farm activities increased the probability to choose smallscale traders. Other factors such as AER I reduced the probability of choosing the other buyers.

The study concludes that socio-economic (household's adult equivalent, household head's age, education level, area planted, quantity of beans harvested, Simpson Index of Diversity), institutional (access to price information and access to extension services, distance to the market), and location factors (AER I, AER III) are important factors in determining market participation, extent of participation and channel choice decisions among smallholder mixed bean farmers in Zambia. Based on these findings, policy recommendations can be made to increase market participation, the extent of participation and improve access to lucrative market outlets.

5.3 Recommendations

The small-holder bean producers in Zambia have the potential to graduate into commercial farming and reduce poverty levels in the rural areas. This can be achieved through full participation in the bean market.

Deleted: ¶

Based on the study results, the following recommendations or policy implications are drawn in order to address some of the factors constraining full participation. Crop diversification should be encouraged through tailored extension messages. This and an NGO and government sponsored crop diversification programme are a necessity in achieving crop diversity. Farmers should be encouraged to increase the area allocated to mixed beans to ensure surplus production and market participation. The need to increase mixed beans farmers' access to price information is cardinal to enable timely informed decision-making in terms of choice of marketing channel outlets. This should be complimented by infrastructure development in terms of good road network to enhance market participation and movement of produce to lucrative market outlets. Good road network is expected to reduce dependence on limited market channels and on the costs associated with poor roads.

5.4. Limitations of the study

The study used secondary data collected by IAPRI/CSO. The data collected for the overall sample was broad and not entirely focused on the crop of interest (mixed beans). However, despite the limitations, findings of the study on factors influencing market participation and marketing channel choice decisions can be generalized.

Deleted: ¶

REFERENCES

- Abafita, J., Atkinson, J. and Kim, C.S. 2016. Smallholder commercialization in Ethiopia: Market orientation and participation. *International Food Research Journal* 23, 1797–1807.
- Abayneh, Y. and Tefera, T. 2013. Factors influencing market participation decision and extent of participation of haricot bean farmers in Meskan District, Ethiopia. *International Journal of Management and Development Studies*. Vol 2, Issue. 8: 2320-0685.
- Abbott, C. 1987. *Agricultural marketing for the developing world*. Cambridge University Press, UK.
- Achike, A. I. and Anzaku, T. A. K. 2010. Economic analysis of the marketing margin of benniseed in nasarawa state, Nigeria. *Journal of Tropical Agriculture, Food, Environment and Extension*. 9 (1): pp. 47 - 55
- Adenegan K. O., Adepoju, A. and Nwauwa, L.O.E. 2012. Determinants of market participation of maize farmers in rural Osun State of Nigeria. *International Journal* of Agricultural Economics & Rural Development, 5 (1)
- Adeyokunnu, T. O. 1973. Marketing Margins for Rice in Egba Division of Western Nigeria. Bulletin of Rural Economics and Sociology. 8:243-253. ISSN 0007-5078.
- Afriseed. 2018. Beans Kabulangeti. [Available at]: <u>http://www.stewardsglobe.com/2018/04/30/beans-kabulangeti/</u> Accessed: 19/11/2019
- Amao, A.I. and Egbetokun, O. A. 2018. Market participation among vegetable farmers, International Journal of Vegetable Science, 24:1, 3-9,
- <u>Anh</u>, N. H. and Bokelmann, W. 2019. Determinants of smallholders' market preferences: The case of sustainable certified coffee farmers in Vietnam. *Sustainability*, 11(10), 2897; a<u>https://doi.org/10.3390/su11102897</u>

- Anteneh. A., Muradian, R. and Ruben, R. 2011. Factors affecting coffee farmers market outlet choice in Ethiopia. A paper prepared for the EMNet 2011 in Cyprus (December 1-3). Centre for International Development Issues Nijmegen: Radboud University, the Netherlands.
- Apind, B. O. 2015. Determinants of Smallholder Farmers Market Participation; A Case Study of Rice Marketing in Ahero Irrigation Scheme. An MSc Thesis Submitted to the Egerton Graduate School.
- Arinloye, D-D. A. A., Pascucci, S., Linnemann, A. R., Coulibaly, O. N., Hagelaar, G., and Omta, O.S.W.F. 2015. Marketing channel selection by smallholder farmers. *Journal of Food Products and Marketing*. 21 (4), 337–357.
- Asfaw, S., Lipper, L., Dalton, T. J. and Audi, P. 2012. "Market participation, on-farm crop diversity and household welfare: micro-evidence from Kenya." *Journal of Environment and Development Economics*. UK: Cambridge University Press. Pp 579-601
- Assefa, A., Tana, T., Nigussie, D. and Dessalegn, Y. 2017. Maize-common bean intercropping to optimize maize-based crop production. *The Journal of Agricultural Science*, 155(7):1-13
- Awotide, B. A., Diagne, A., and Awoyemi, T. T. 2013. Agricultural Technology Adoption, Market Participation and Rural Farming Households' Welfare in Nigeria. Conference Presentation.
- Bahta, S., and S. Bauer. 2012. Policy options for improving market participation and sales of smallholder crop producers: A case study of the Free State Province of South Africa. African Journal of Agricultural Research 7(24):3525-3533
- Barrett C. B. 2008. Smallholder market participation: Concepts and evidence from Eastern and Southern Africa. *Food Policy*, Vol. 33: p299-317.

- Barrett, C.B., Marenya, P.P., Mc Peak, J.G., Minten, B., Murithi., F.M. and Wangila, J. 2006. Welfare Dynamics in Rural Kenya and Madagascar. *Journal of Development Studies*, 42:248-277.
- Beadgie, W., and Reddy, P. 2020. Determinants of Market Participation and Intensity of Marketed Surplus Among Maize Producers in Fogera District of South Gondar Zone, Ethiopia: Heckman Two Stage Equation Approach. <u>DOI: 10.21203/rs.3.rs-132809/v1</u>
- Bellemare, M. F. and Barrett C. F. 2006. An ordered tobit model of market participation: Evidence from Kenya and Ethiopia. *American Journal of Agricultural Economics*. 88(2), 324-337.
- Bongiwe, G. and Masuku, B. 2012. Factors affecting the choice of marketing channel by vegetable farmers in Swaziland. Canadian Center of Science and Education. *Sustainable Agriculture Research*, 2 (1):123.
- Boughton, D., Mather, D., Barret, C. B., Benfica, R., Abdula, D., Tschirley, D. and Cunguara, B., 2007. "Market participation by rural households in a low-income country: an asset-based approach applied to Mozambique". *Journal on Faith and Economics*, 50: 64 – 101
- Central Statistical Office (CSO). 2010. Census of Population and Housing: National Analytical Report. Lusaka: Central Statistical Office.
- Certo, S. T., Busenbark, J. R., Woo, H. S., and Semadeni, M., 2016. Sample selection bias and heckman models in strategic management research. *Strategic Management Journal*, 37: 2639–2657 DOI: 10.1002/smj.2475
- Chalwe, S. 2011. Factors influencing bean producers' choice of marketing channels in Zambia. Lusaka: University of Zambia.
- Cheruiyot, E.K., Mumera, L.M., Nakhone, L.N. and Mwonga, S.M. 2001. Rotational effects of grain legumes on maize performance in the rift valley highlands of Kenya. *African Crop Science Journal, Vol. 9. No. 4, pp. 667-676*

- Chirwa, E. W. 2009. Determinants of Marketing Channels among Smallholder Maize Farmers in Malawi. University of Malawi - Chancellor College, Working Paper No. 2009/03
- Delgado, C. L. 1999. Sources of growth in smallholder agriculture in sub-Saharan Africa: the role of vertical integration of smallholders with processors and marketers of high value-added items. *Agrekon*, 38(Special issue):165-189.
- Dessie, A.B., Abate, T.M. and Mekie M.M. 2018. Factors affecting market outlet choice of wheat producers in North Gondar Zone, Ethiopia. *Journal of Agriculture and Food Security*, 7:91.
- Doss, C. 2001. Designing Agricultural Technology for African Women Farmers: Lessons from 25 Years of Experience', *World Development*, 29 (12), pp. 2075-2092.
- Dougherty, C. 1992. Introduction to Econometrics. Oxford University Press, New York.
- Ekoja, O. G., Eneh, C. N., Ugba, A. V. and Ogaga, A. I. 2018. Economic analysis of beans production for marketing profitability in north-west Nigeria. *Global Journal of Engineering Science and Research Management*, 5 (5) ISSN 2349-4506
- FAO, IFAD, UNICEF, WFP and WHO. 2018. The State of Food Security and Nutrition in the World 2018: Building climate resilience for food security and nutrition, Rome: FAO.
- Faris A., Shumeta, Z. and Muche, M. 2018. Determinants of Marketed Surplus of Potato Producers in Dedo District of Jimma Zone, Ethiopia. World Journal of Agricultural Sciences 14 (2): 74-80. DOI: 10.5829/idosi.wjas.2018.74.80
- Ferto, I. and Szabo, G. G, 2002. The choice of supply channels in Hungarian fruit and vegetable Sector. Paper for Presentation to the American Agricultural Economics Association Annual Meeting, Long Beach, California.
- Fischer, E. and Qaim, M. 2012. Gender, agricultural commercialization, and collective action in Kenya, Food Security, 4, pp. 441–453.
- Food and Agricultural Organization (FAO). 1994. *Women, agriculture and rural development: A synthesis report of the Africa region.* Rome: Food and Agriculture Organization of the United Nations.
- Food and Agriculture Organization of the United Nations. 2015. The economic lives of smallholder farmers: An analysis based on household data from nine countries. Rome: FAO/UN.
- Food and Agriculture Organization of the United Nations. 2018. Cropping system diversification in Eastern and Southern Africa: Identifying policy options to enhance productivity and build resilience. FAO Agricultural Development Economics Working Paper 18-05
- Funke, T. B. 2006. From farm to retail: Costs and margins for selected food industries in South Africa. An M.Sc. Thesis in Agricultural Economics. Faculty of Economic and Management Studies, University of Pretoria. South Africa.
- Gebremedhin, B and Hoekstra, D. 2008. "Cereal Marketing and Household Market Participation in Ethiopia: The Case of Teff, Wheat and Rice," Second International Conference, Accra, Ghana 52102, African Association of Agricultural Economists (AAAE).
- Geil, P. B., and Anderson, J. W. 2013. Nutrition and health implications of dry beans: a review. *Journal of the American College of Nutrition*. 13 (6): pp 549 – 558.
- Gido, E. O., Ayuya, O. I., Owuor, G., and Bokelmann, W. 2016. Consumer's choice of retail outlets for African indigenous vegetables: Emperical evidence among rural and urban households in Kenya. *Cogent Food and Agriculture*, 7, 1-14 <u>http://doi.org/10.1080/2311932.2016.1248523</u>
- Goetz., S. J. 1992. A selectivity model of household food marketing behavior in Sub-Saharan Africa. American Journal of Agricultural Economics. Vol. 74, No. 2 pp. 444-452

- Government of the Republic of Zambia. 2013. Zambia National Agriculture Investment Plan (NAIP) 2014-2018 under the Comprehensive Africa Agriculture Development Programme (CAADP). Ministry of Agriculture and Livestock.
- Greene, W.H. 2000. Econometric analysis. Fourth edition, Prentice Hall International, Inc., International edition, U.S.A. 1004p.
- Haggblade & Tembo. 2003. Conservation Farming in Zambia (A paper prepared for the International Workshop on "Reconciling Rural Poverty and Resource Conservation: Identifying Relationships and Remedies), Cornell University, Ithaca, New York
- Hailu, C. and Fana, C. 2017. Determinants of market outlet choice for major vegetables crop: evidence from smallholder farmers of Ambo and Toke-Kutaye Districts, West Shewa, Ethiopia. *International Journal of Agricultural Marketing, Vol. 4(2),* pp. 161-169
- Hamazakaza, P; Katungi. E.; Ryes, B.; Maredia, M.; Muimui, K.; Ojara, M. 2014. Assessing access and adoption of common bean improved varieties in Zambia . Zambia Agriculture Research Institute (ZARI); Centro Internacional de Agricultura Tropical (CIAT); Michigan State University, Zambia. 42 p. (Research Technical Report).
- Han, C. 2011. Obstacles to development of marketing channels of agricultural products in China and countermeasures. Asian Agricultural Research, (3) 7
- Heckman, J. 1979. Sample selection bias as a specification error, *Econometrica*, 47, pp. 153-61
- Heltberg, R., and F. Tarp. 2002. "Agricultural supply response and poverty in Mozambique." Food Policy 27:103–124
- Hichaambwa, M. and Mofya-Mukuka, R. 2016. What Factors Drive Smallholder Crop Diversification in Zambia? INDABA Policy Brief No. 82

- Hichaambwa, M., Beaver, M., Chapoto, A., and Weber, M. 2009. Patterns of Urban Food Consumption and Expenditure in Zambia: An Overview Report Based on the CSO/MACO/FSRP Food Consumption Survey in Urban Areas of Lusaka, Kitwe, Mansa and Kasama, 2007-2008. Working paper No. 43. Food Security Research Project. Lusaka, Zambia.
- Hoffman, R. and Kassouf, A.L. 2005. Deriving conditional and unconditional marginal effects in log earning equations estimated by Heckman's procedure. *Applied Economics, Londres*, 37:829-844
- International Labour Organization (ILO). 2009. *Global employment trends for women*. Geneva: International Labour Organization.
- Indaba Agricultural Policy Research Institute (IAPRI). 2016. Rural Agricultural Livelihoods Survey 2015 Survey Report. Lusaka: IAPRI
- Jagwe, J. N. 2011. Impact of Transaction Costs on the Participation of Smallholder Farmers and Intermediaries in the Banana Markets of Burundi, Democratic Republic of Congo and Rwanda. Ph.D. Thesis (Agricultural Economics), University of Pretoria, Pretoria, submitted.
- Jagwe, J., Machethe, C. L. and Ouma, E. 2010. Transaction costs and smallholder farmers' participation in banana markets in the Great Lakes Region of Burundi, Rwanda and the Democratic Republic of Congo. *The African Journal of Agricultural and Resource Economics, Vol 6 No 1*
- Jari, B and Fraser, C.G. 2009. An analysis of institutional and technical factors influencing agricultural marketing amongst smallholder farmers in the Kat river valley, Eastern Cape, South Africa. *African Journal of Agricultural Research*, 4 (11): 1129-1137.
- Kaba, S. D. 2016. Agricultural Technology Adoption, Crop Diversification and Efficiency of Maize-Dominated Smallholder Farming System in Jimma Zone, South-Western Ethiopia. A Phd Dissertation submitted to Haramaya University, Haramaya.

- Kabeto, A. J. (2014). An analysis of factors influencing participation of smallholder farmers in red bean marketing in Halaba Special District, Ethiopia. An MSc Thesis submitted to the Faculty of Agriculture, University of Nairobi.
- Kamaraa, A., Conteh, A., Rhodes, E.R. and Cooke, R.A. 2019. The relevance of smallholder farming to African agricultural growth and development. *African Journal of Food, Agriculture, Nutrition and Development*, 19, (1) DOI: 10.18697/ajfand.84.BLFB1010
- Kamfwa, K. 2016. Racing for Beans: How Breeding Improved Local Bean Varieties Accelerates Food and Nutritional Security in Zambia. Agrilinks/Feed the Future. Available online at: <u>https://www.agrilinks.org/blog/racing-beans-how-breeding-improved-local-bean-varieties-accelerates-food-and-nutritional</u> 22/05/2020
- Kamunye, K. K. 2016. An analysis of determinants of market participation among smallholder common bean farmers in Rwanda, a gendered approach. An MSc thesis submitted to the Graduate School of Economics of Egerton University.
- Kamunye, K. K., Mshenga, P. M., and Birachi, E. A. 2016. Determinants of smallholder common bean market participation and extent of participation in Rwanda by gender. Pan-African Grain Legume and World Cowpea Conference Paper, Livingstone, Zambia.
- Katungi, E., Farrow A., Chianu, J., Sperling, L., and Beebe, S. 2009. Common bean in Eastern and Southern Africa: a situation and outlook analysis. International Centre for Tropical Agriculture.
- Katungi, E., Karanja D., Wozemba, D., Mutuoki, T., and Rubyogo, J.C.,2011. A costbenefit analysis of farmer based seed production for common bean in Kenya. *African Crop Science Journal*, 19 (4), 409- 415.

- Khalil, C.A., Conforti, P., Ergin, I., and Gennari, P. 2017. Defining small scale food producers to monitor target 2.3. of the 2030 agenda for sustainable development. Working Paper Series ESS/17-12. Rome: FAO/UN.
- Kihoro, E. M. 2016. An analysis of factors influencing farmers' choice of Greengram marketing channels in Mbeere South Sub-County, Kenya. An MSc thesis submitted to the University of Nairobi.
- Kimei, S., Mshenga, P., and Birachi, E. 2017. Factors influencing commercialization of beans among smallholder farmers in Rwanda, *Journal of Agriculture and Veterinary Science (IOSR-JAVS) Volume 10, Issue 8.*
- Kirui, H., Mutai, K. B., Kibet K. L., Kibet R. P., and Macharia, A. 2016. Determinants of tea marketing channel choice and sales intensity among smallholder farmers in Kericho District, Kenya. *Journal of Economics and Sustainable Development Vol.7, No.7.*
- Komarek, A. 2010. The determinants of banana market commercialization in Western Uganda. *African Journal of Agricultural Research*, 5(9): 775-784.
- Kuma, B., Baker D., Getnet, K. and Kassa, B. 2014. Factors affecting milk market participation and volume of supply in Ethiopia. Asian Journal of Rural Development. 4 (1): 1-15.
- Kuma, B., Baker, D., Getnet, K. and Kassa, B. 2013. Factors affecting milk market outlet choices in Wolaita Zone, Ethiopia. *African Journal of Agricultural Research*. Vol. 8(21), pp. 2493-2501. DOI: 10.5897/AJAR11.2156
- Kumar, P., Kannan, Chaudhary, R., and Vishnu, K. 2013. Assessment of marketed and marketable surplus of food-grain crops in Karnataka. Agricultural Development and Rural Transformation Centre: Banglore New Dehli
- Kumaraswamy, S.K., Mattern, M., and Hernandez, E. 2020. Assets Matter to Poor People: But What Do We Know about Financing Assets? Working Paper. Consultative Group to Assist the Poor. World Bank

- Kyaw, N. N., Ahn, S. and Lee, S. H. 2018. Analysis of the factors influencing market participation among smallholder rice farmers in Magway Region, Central Dry Zone of Myanmar, *Sustainability*, *10*, *4441*; doi:10.3390/su10124441
- Latt, E. A and Nieuwoudt W. L. 1988. Identification of plot size effects on commercialization of small-scale agriculture in Kwazulu. *Development Southern Africa*, Vol. 5 (3): p371-382.
- Mabuza, M.L., Wale, E., and Ortmann, G.F. 2014. Effects of transaction costs on mushroom producers' choice of marketing channels: Implications for access to agricultural markets in Swaziland. <u>South African Journal of Economic and</u> <u>Management Sciences</u>, 17(2):207-219. DOI: <u>10.4102/sajems.v17i2.494</u>
- Machethe, C.L. 2004. Agriculture and poverty in South Africa: Can agriculture reduce poverty? Overcoming Underdevelopment Conference, Pretoria, South Africa, 28– 29 Oct. 2004. (<u>http://sarpn.org/documents/d0001005/P1125-Agriculture-Poverty_Machethe_2004.pdf</u>).
- Maddala, G. S.1983 Limited-Dependent and Quantitative Variable in Econometric. (1st edition). Cambridge University Press, Cambridge.
- Maina, C. M., Lagat, J. K., and Mutai, B .K. 2015. Effect of transaction costs on choice of mango marketing channel: the case of small scale farmers in Makueni County, Kenya. *Journal of Agriculture and Veterinary Science*, 8 (4), 54-62.
- Makhura M. T. 2001 (unpublished PhD Dissertation). Overcoming transaction costs barriers to market participation of smallholder farmers in the Northern Province of South Africa.
- Makhura M. T., Goode M. F. and Coetzee G. K. 1997. Indexing participation in the market economy through factor analysis: Implications for food security, *Agrekon*, Vol. 36 (4): p473-483

- Mamo, G and Degnet, A. 2012. Patterns and determinants of livestock farmers' choice of marketing channels: Micro-level Evidence. EEA/EEPRI working paper, Addis Ababa.
- Mamo, G. 2009. Choice of marketing channels and transaction costs: The case of Maize Marketing in Bura Borama, Shashemene District. M.Sc. Thesis Presented to the School of Graduate Studies of Addis Ababa University. 74p.
- Mango, N., Makate, C., Francesconi, N., Jager, M., and Lundy, M. 2018. Determinants of market participation and marketing channels in smallholder groundnut farming: A case of Mudzi district, Zimbabwe, *African Journal of Science, Technology, Innovation and Development*, 10 (3), 311-321. DOI: 10.1080/20421338.2018.1457274
- Marenya, P.P., and Barret. B. 2006. Household-level determinants of adoption of improved natural resources management practices among smallholder farmers in Western Kenya. *Food Policy*, 32 (1):151-536.
- Masuku M. B., Makhura M. T. and Rwelarmira J. K. 2001. Factors affecting marketing decisions in the maize supply chain among smallholders in Swaziland. Agrekon: *Agricultural Economics Research, Policy and Practice in South Africa*, 40(4): 698-707.
- Mathenge, M., Frank, P., John, O., and Dagmar, M. 2010 Participation in agricultural markets among the Poor and marginalized: analysis of factors Influencing participation and impacts on income and Poverty in Kenya, Study report, *world agro forestry center*.
- Mather, D., Boughton, D., and Jayne, T.S. 2011. Smallholder Heterogeneity and Maize Market Participation in Southern and Eastern Africa: Implications for Investment Strategies to Increase Marketed Food Staple Supply: MSU International Development, Working Paper.

- Mather, D., Boughton, D., and Jayne, T.S. 2013. Explaining smallholder maize marketing in southern and eastern Africa: The roles of market access, technology and household resource endowments. *Food Policy* 43 (2013) 248-266.
- Mbitsemunda, J.P.K., and Karangwa, A. 2017. Analysis of factors influencing market participation of smallholder bean farmers in Nyanza District of Southern Province, Rwanda. *Journal of Agricultural Science;* Vol. 9, No. 11.
- McFadden, D. 1974. The measurement of urban travel demand. *Journal of Public Economics*, 3: 303–328.
- McNamara, P. and Tata, J. 2015. "Principles of designing and implementing Agricultural Extension Programs for Reducing Post-Harvest Loss", *Agriculture*, Vol. 5 No. 4, pp. 1035-1046.
- Mendoza, G. 1995. A primer on marketing channels and margins. In G. J. Scott (eds.). Prices, Products, and People: Analyzing Agricultural Markets in Developing Countries. Lynne Reinner Publishers, Boulder, London. pp. 257-275.
- Messina, V. 2014. Nutritional and health benefits of dried beans. *The American Journal* of Clinical Nutrition. 100 (1): pp 437S–442S.
- Ministry of Agriculture (MA) and Ministry of Fisheries and Livestock (MFL). 2016. Second National Agricultural Policy. Lusaka: GRZ
- Ministry of Agriculture and the Central Statistical Office. 2017. 2016/2017 Crop Forecast Survey Results Presentation. Lusaka: GRZ
- Mirie, T. and Zemedu, L. 2018. Determinants of market participation and intensity of marketed surplus among teff producers in Dera District of South Gondar Zone, Ethiopia. Journal of Development and Agricultural Economics, Vol. 10(10), pp. 359-366

- Mmbando, F.E. 2014. Market participation, channel choice and impacts on household welfare: the case of smallholder farmers in Tanzania. A PhD Thesis submitted to the Graduate School of the University of KwaZulu-Natal, South Africa.
- Mmbando, F. E., Wale, E., Baiyegunhi, L. J. S., and Darroch, M. A. G. 2016. The choice of marketing channel by maize and pigeonpea smallholder farmers: Evidence from the northern and eastern zones of Tanzania. *Agrekon*, Vol 55, Issue 3 p 254-277.
- Mofya-Mukuka, R., Banda, A. and Kabwela, B. 2018. Understanding the Hunger and Malnutrition Situation in Zambia, Working Paper 148, Lusaka: IAPRI.
- Moyo, T. 2010 Determinants of participation of smallholder farmers in the marketing of small grains and strategies for improving their participation in the Limpopo River Basin of Zimbabwe. An MSc. Thesis Submitted to the University of Pretoria
- Muhammed, U. 2011. Market chain analysis of teff and wheat production in Halaba Special Woreda, southern Ethiopia. An MSc. thesis submitted to the School of Graduate Studies, Haramaya University. 104p.
- Muimui, K. K. 2010. Opportunities and challenges. Bean Stakeholders Consultative Workshop. Lusaka, Zambia
- Mukarumbwa, P., Mushunje, A., Taruvinga, A., Akinyemi, B. and Ngarava, S. 2018. Analysis of factors that influence market channel choice of smallholder vegetable farmers in Mashonaland east province of Zimbabwe, *International Journal of Development and Sustainability*, Vol. 7, No. 2: Pages 734-754.
- Mukutu, N., Kalaba, E.C.J., Kidane, P., and Kolshus, H. 2004. Zambia: Support to NEPAD–CAADP Implementation. Lusaka: GRZ/FAO
- Musara, J. P., Musemwa, L., Mutenje, M., Mushunje, A. and Pfukwa, C. 2018. Market participation and marketing channel references by small scale sorghum farmers in semi-arid Zimbabwe. *Agrekon*, Vol 57, Issue 1, p 64-77.
- Mutayoba, V. and Ngaruko, D. 2015. Determinants of farmers' participation in high value crops in Tanzania, *African Journal of Economic Review, Volume III, Issue 2.*

- Muthini, D. N. 2015. An assessment of mango farmers' choice of marketing channels in Makueni, Kenya. MSc Thesis Submitted to The University of Nairobi.
- Mwanamwenge, M. and Harris, J. 2017. Agriculture, Food Systems, Diets and Nutrition in Zambia, Discussion Paper, Lusaka: IIED/Hivos.
- Nedumaran, S., Abinaya, P., Jyosthnaa, P., Shraavya, B., Rao, P. and Bantilan, C. 2015. Grain Legumes Production, Consumption and Trade Trends in Developing Countries. Working Paper, Series No 60. ICRISAT Research Program, Markets, Institutions and Policies. Patancheru 502 324, Telangana, India: International Crops Research Institute for the Semi-Arid Tropics. 64 pp.
- Negeri, M. A. 2017. Determinants of market outlet choice of coffee producing farmers in Lalo Assabi District, West Wollege Zone, Ethiopia: An econometric approach. *Journal of Economics and Development, Vol.19, No.2, pp. 48-67.*
- Ngoh, B. S., Endeley, B. J., Emmanuel, A. T., and Lytia, E. M. 2017. Understanding the production and sales determinants of haricot beans (phaseolus vulgaris l.) in the western highlands of Cameroon. *International Journal of Development Research Vol. 07, Issue, 09, pp.15126-15133*
- Njuki, J., Kaaria, S, Sanginga, P, Kaganzi, E. and Magombo, T. 2009. *Community Agro Enterprise Development, Experiences from Uganda and Malawi*, in Scoones and Thompson.
- Nkwasibwe A, Mugisha J, Elepu G and Kaneene, J. B. 2015. Increasing the efficiency of the dairy value chain in Uganda: Determinants of choice of milk marketing channels by dairy farmers in Kiruhura District, Uganda. *Livestock Research for Rural Development. Volume 27, Article #168.*
- Ogunleye, K. Y., and Oladegi, J. O. 2007. Choice of cocoa market channels among cocoa farmers in ILA local government area of Osun State, Nigeria Department of Agricultural Economics and Extension, Faculty of Agricultural Sciences. *Middle East Journal of Scientific Research*, 2(1), 14-20.

- Ohen, S. B., Etuk, E. A. and Onoja J.A. 2013. Analysis of market participation by rice farmers in southern Nigeria. *Journal of Economics and Sustainable Development*, Vol.4, No.7.
- Okonya, J. S., Mudege, N. N., Rietveld, A. M., Nduwayezu, A., Kantungeko, D., Hakizimana, B. M. Nyaga, J. N., Blomme, G., Legg, J. P. and Kroschel, J. 2019. The role of women in production and management of RTB crops in Rwanda and Burundi: Do men decide, and women work? *Journal of Sustainability*, Vol 11, Issue 4304
- Omiti, J. M., Otieno, D. J., Nanyamba, T. O., and McCullough, E. 2009. Factors influencing the intensity of market participation by smallholder farmers: A case study of rural and peri-urban areas of Kenya. *The African Journal of Agricultural* and Resource Economic, Vol 3 No. 1
- Onyeabor, E. N. 2009. Marketing and Distribution Channels; Agricultural Marketing for Developing Economies. Enugu John Jacob's Classic Publishers Ltd; 14-25
- Osmani, A. G. and Hossain, E. 2015. Market participation decision of smallholder farmer and its determinants in Bangladesh. *Economics of Agriculture Journal*. (549.3).
- Otekunrin, O. A., Momoh, S., and Ayinde, I. A. 2019. Smallholder farmers' market participation: concepts and methodological approaches from Sub-Saharan Africa. *Current Agriculture Research Journal*, Vol. 7, No.(2) 2019, pg. 139-157
- Pace, J. D. 2012. Factors affecting cotton producers' choice of marketing outlet. An MSc. Thesis Submitted to the Office of Graduate Studies of Texas A & M University.
- Pan African Bean Regional Alliance (PABRA). 2015 'The Promise of Beans' Available online at: <u>http://www.pabra-africa.org/the-promise-of-beans/</u> Accessed: 15/11/2019
- Ponte, S. 2002. The late revolution regulation, markets and consumption in the global coffee chain. *World Development*, 30 (7): 1099-1122.

- Porter, M. and Miller, V.E. 1985. How information gives you competitive advantage. Harvard Business Review. p. 149-160
- Qadri, B. 2018. Role of Marketing Channels. Kashmir: Central University of Kashmir
- Rahman, I. H., Envulus, E.E. and Oyewole, S.O. 2009. Income and crop diversification among farming households in a rural area of north central Nigeria. Journal of Tropical Agriculture, Food, Environment and Extension. Vol 8 No. 2 pp 84 -89
- Rehima, M. 2006. Analysis of red pepper marketing: the case of Alaba and Siltie in SNNPRS of Ethiopia. An M.Sc Thesis Presented to the School of Graduate Studies of Hararnaya University.
- Reyes, B., Donovan, C., Bernsten R. and Maredia, M. 2012. Market participation and sale of potatoes by smallholder farmers in the central highlands of Angola: A double hurdle approach. Michigan State University, USA.
- Riziki, J., M., Mlongo, P., M., Mwanarusi, S., Agnes, N. and Ipomai, S. 2015. Determinants of choice of marketing outlets for African indigenous vegetables among the agro-pastoral Maasai of Narok and Kajiado counties of Kenya, *Journal* of Economics and Sustainable Development, 8 (6), 29-42.
- Sadoulet, E.; De Janvry, A. 1995. Quantitative Development Policy Analysis: Exercise Solutions; Johns Hopkins University Press: Baltimore, MD
- Salmerón-Miranda, F. 2008. Nitrogen Use in a Maize-Bean rotation in Nicaragua Effects of organic and mineral fertilizers. A PhD Thesis submitted to the Swedish University of Agricultural Sciences Uppsala.
- Sartori, A. E. 2003. An estimator for some binary-outcome selection models without exclusion restrictions. *Political Analysis*, 11(2): 111–138.
- Scott, G.J. 1995. Prices, Products and People: Analyzing Agricultural Markets in Developing Countries. London: Lynne Reinner Publishers, Boulder. p498.

- Sebatta, C., Mugisha, J., Katungi, E., Kashaaru, A., and Kyomugisha, H. 2014. Smallholder farmers' decision and level of participation in the potato market in Uganda. *Modern Economy*, 5, 895-906.
- Shiferaw, B., Obare, G., and Muricho, G. 2006. Rural institutions and producer organizations in imperfect markets: Experience from producer marketing groups in semi-arid eastern Kenya. Socioeconomics and Policy Working Paper Series no. 23
- Shilpi, F. and Umali-Deininger, D, 2008. Market facilities and agricultural marketing: Evidence from Tamil Nadu, India. *Journal of Agricultural Economics*, 39: 281-294.
- Siame J., Willey, R.W. and Morse S. 1998. The response of maize/*Phaseolus* intercropping to applied nitrogen on Oxisols in northern Zambia. *Field Crops Research*, (55, 1-2) pp 73-81.
- Siegel, P.B. and Alwang, J. 2005. Poverty Reducing Potential of Smallholder Agriculture in Zambia: Opportunities and Constraints. Africa Region Working Paper Series No. 85. Washington DC: World Bank.
- Sichilima, T., Mapemba, L. and Tembo, G. 2016. Drivers of dry common beans trade in Lusaka, Zambia: A Trader's Perspective. Sustainable Agriculture Research; Vol. 5, No. 2
- Siddique, M. I., Garnevska, E. and Marr, N. E. 2018. "Factors affecting marketing channel choice decisions of smallholder Citrus growers", *Journal of Agribusiness in Developing and Emerging Economies*, Vol. 8 Issue: 3, pp.426-453,
- Sigei, G. 2014. Determinants of market participation among small-scale pineapple farmers in Kericho County, Kenya. MSc Thesis Submitted to the Graduate School of Egerton University, Kenya
- Sigei, K. G., Bett, K. H., Kibet, K. L. and Mutai C. M. 2013. Determinants of market participation among small-scale pineapple farmers in Kericho County, Kenya. *Journal of Economics and Sustainable Development, Vol.4, No.19*

- Singh, D. K. 2018. Determinants of dairy farmers' choice of marketing channels in Bihar, India. Agricultural Economics Research Review, 31 (1), 149-155
- Siri, B. N., S. B., Ojong, Tabe, M. P., and Molua, E. L. 2018. Socioeconomics of the production and marketing of haricot beans (phaseolus vulgaris 1.) in the western highlands of Cameroon. *Journal of Economics and Sustainable Development*, Vol.9, No.8.
- Siziba, S., Nyikahadzoi, K., Diagne, A., Fatunbi A. O., and Adekunle, A. A. 2011. Determinants of cereal market participation by sub-Saharan Africa smallholder farmer. *Journal of Agriculture and Environmental Studies*, vol. 2, no. 1, pp. 180-193
- Takele, A. 2010. Analysis of Rice Profitability and Marketing Chain: The case of Fogera Woreda, South Gondar Zone, Amhara National Regional State, Ethiopia: MSc Thesis (Agricultural Economics). Haramaya University:
- Tefera, T. 2014. Determinants of smallholder pulse producers market orientation in Southern Ethiopia. *Asian Journal of Business Management* 6(2): 97-103
- Tembo, S. and Sitko, N. 2013. Technical Compendium: Descriptive Agricultural Statistical Analysis for Zambia. Working Paper. Lusaka: IAPRI.
- Tembo, S., & Sitko, N. 2013. Technical Compendium: Descriptive Agricultural Statistics and Analysis for Zambia. Working Paper 76. Indaba Agricultural Policy Research Institute (IAPRI) Lusaka, Zambia.
- Tesfaw A. 2013. Determinants of agricultural commodity market supply: a case study in the upper watershed of the Blue Nile, Northwestern Ethiopia. *Journal of Agribusiness and Rural Development*. 04 (30): pp.243-256.
- Timmer, P. C. 1988. The agricultural transformation. Handbook of Development Economics. Vol 1, Vol I.

- Tola, M. 2014. Tomato value chain analysis in the central rift valley: The case of Dugda woreda, East Shoa Zone, Oromia National Region State, Ethiopia. A Master of Science Thesis dissertation submitted to Haramaya University, Haramaya.
- Tsourgiannis, L., Errington, A., and Eddison, J. 2002. "Marketing Strategies of Agricultural Producers in Objective One Greek Regions: The Factors Affecting the Selection of Marketing Channels of Sheep and Goat Producers" School of Geography, University of Plymouth, Drake Circus, Plymouth, UK
- Tumukunde, E.S. 2018. Determinants of choice of marketing channels among potato farmers in Musanje District, Rwanda: Evidence after the 2015 potato market reforms. MSc. Thesis submitted to University of Nairobi.
- Twumasi, M. A., Jiang, Y., and Acheampong, M. O. 2019. Determinants of agriculture participation among tertiary institution youths in Ghana. *Journal of Agricultural Extension and Rural Development*, 11(3), 56-66.
- Urquieta, A. 2009. Effects of access to information on farmer's market channel choice: The case of potato in Tiraque Sub-Watershed (Cochabamba - Bolivia). MSc thesis submitted to the Faculty of the Virginia Polytechnic Institute and State University. Blacksburg, Virginia.102p.
- Von Braun, J., De Haen, H. and Blanken, J. 1991. Commercialization of agriculture underpopulation pressure: Effects on production, consumption and nutrition in Rwanda. Research report 85, International Food Policy Research Institute (IFPRI).
- Von Braun, J. 1995. Agricultural commercialization and diversification: processes and policies, *Food Policy. Vol 20, Issue 3 pp 171-185*.
- Von Braun, J., Puetz, D., and Webb, P. 1989. Irrigation technology and commercialization of rice in the Gambia: effects on income and nutrition. Research Report. IFPRI
- Welthungerhilfe (WHH) and Concern Worldwide (CWW). 2018. 2018 Global Hunger Index: Forced migration and hunger, Bonn and Dublin: Welthungerhilfe and Concern Worldwide.

- Willams, R. 2020. Heteroskedasticity. University of Notre Dame. [Available online at]: https://www3.nd.edu/~rwilliam/stats2/125.pdf. Accessed on 30/04/2020.
- Wolday A. 1994. Food grain marketing development in Ethiopia after reform 1990. A case study of Alaba Siraro. The PhD Dissertation Presented to Verlag Koster University. Berlin 293p.
- Wooldridge, J. M. 2008. Basic Econometrics, 2nd Edition. Massachusetts Institute of Technology.
- Wooldridge, J. M. 2012. Introductory Econometrics, A Modern Approach, 5th Edition.
- World Bank, Food and Agriculture Organization and International Fund for Agricultural Development. 2009. Gender in Agriculture: Sourcebook. Washington DC: World Bank.
- Zamasiya, B., Mango, N., Nyikahadzoi, K. and Siziba, S. 2014. "Determinants of soybean market participation by smallholder farmers in Zimbabwe", *Journal of Development and Agricultural Economics*, Vol. 6 No. 2, pp. 49- 58.

APPENDICES

Appendix 1. Variance Inflation Factor (VIF) results for the Heckman model

| . vif | | |
|--|--|--|
| Variable | VIF | 1/VIF |
| mixedbeansha totharv AERIII edu_max AERI TLU ae net_off_farm dhhsex dcoop mem | 1.50 1.46 1.39 1.21 1.17 1.17 1.17 1.12 1.11 1.11 | 0.666089 0.683454 0.720262 0.823771 0.825616 0.852529 0.857273 0.891457 0.897863 0.899408 |
| dadv_ext SID_crops headage dloan landholdsz15 dinf_price dist_market | 1.11 1.10 1.06 1.05 1.05 1.02 | 0.903203 0.908711 0.911071 0.946886 0.947935 0.952846 0.976033 |
| Mean VIF | 1.17 | |

Appendix 2, Breusch-Pagan / Cook-Weisberg test for heteroskedasticity – Heckman Model

. hettest headage dhhsex a
e du_max landholdsz15 mixedbeansha totharv dloan SID_crop
 > s dinf_price dloan dcoop_mem_net_off_farm dist_market dadv_ext TLU AERI AERIII

```
Breusch-Pagan / Cook-Weisberg test for heteroskedasticity
Ho: Constant variance
Variables: headage dhhsex ae edu_max landholdsz15 mixedbeansha totharv
dloan SID_crops dinf_price dcoop_mem net_off_farm dist_market
dadv_ext TLU AERI AERIII
chi2(17) = 39.82
```

chi2(17) = 39.82 Prob > chi2 = 0.0014 Formatted: Font: Not Bold Formatted: Heading 2, Left, Line spacing: single Deleted: 2

Appendix 3, Heckman regression results of mixed bean market participation and intensity

of participation.

. heckman kgsold_total headage dhhsex ae edu_max landholdsz15 mixedbeansha totharv d
> loan dcoop_mem SID_crops dinf_price dist_market dadv_ext TLU AERI AERIII, select(t
> yphh = headage dhhsex ae edu_max landholdsz15 mixedbeansha totharv SID_crops dinf_
> price dloan dcoop_mem net_off_farm dist_market dadv_ext TLU AERI AERIII) twostep

| Heckman | selection | model | two-step | estimates | Number | of obs | = | 1,283 |
|----------|------------|----------|------------|-----------|--------|-------------|---|-------|
| (regress | sion model | with sam | ple select | tion) | 5 | Selected | = | 867 |
| | | | | | 1 | lonselected | = | 416 |

| | Nonselect | ted = | 416 |
|------|-----------|-------|----------|
| Wald | chi2(16) | = | 14249.22 |
| Prob | > chi2 | = | 0.0000 |

| | Coef. | Std. Err. | Z | ₽> z | [95% Conf. | Interval] |
|-------------------|-----------|-----------|--------|-------|------------|-----------|
| kgsold total | | | | | | |
| headage | 5501749 | .1870035 | -2.94 | 0.003 | 916695 | 1836548 |
| dhhsex | .5802521 | 6.950908 | 0.08 | 0.933 | -13.04328 | 14.20378 |
| ae | -4.869571 | 1.241768 | -3.92 | 0.000 | -7.303392 | -2.435751 |
| edu max | 2.475904 | .8978239 | 2.76 | 0.006 | .716202 | 4.235607 |
| landholdsz15 | .1865147 | .1647481 | 1.13 | 0.258 | 1363857 | .509415 |
| mixedbeansha | 18.79244 | 4.951497 | 3.80 | 0.000 | 9.087688 | 28.4972 |
| totharv | .7886269 | .0073404 | 107.44 | 0.000 | .7742399 | .8030139 |
| dloan | 1.697414 | 8.712968 | 0.19 | 0.846 | -15.37969 | 18.77452 |
| dcoop mem | -2.945389 | 5.388916 | -0.55 | 0.585 | -13.50747 | 7.616693 |
| SID crops | -6.173098 | 16.34557 | -0.38 | 0.706 | -38.20983 | 25.86363 |
| dinf price | 2.042185 | 5.782466 | 0.35 | 0.724 | -9.291241 | 13.37561 |
| dist market | .0797921 | .0717933 | 1.11 | 0.266 | 0609203 | .2205044 |
| dadv ext | 5.772365 | 4.977633 | 1.16 | 0.246 | -3.983617 | 15.52835 |
| TLU | -1.126634 | .4213863 | -2.67 | 0.008 | -1.952536 | 3007323 |
| AERI | -37.78253 | 20.55902 | -1.84 | 0.066 | -78.07747 | 2.512406 |
| AERIII | -8.424109 | 8.036271 | -1.05 | 0.295 | -24.17491 | 7.326692 |
| _ ^{cons} | -11.2572 | 21.77011 | -0.52 | 0.605 | -53.92583 | 31.41143 |
| typhh | | | | | | |
| headage | 0041198 | .0030272 | -1.36 | 0.174 | 010053 | .0018134 |
| dhhsex | .0423967 | .111394 | 0.38 | 0.703 | 1759314 | .2607249 |
| ae | 0016384 | .0209309 | -0.08 | 0.938 | 0426622 | .0393853 |
| edu max | .024287 | .0147892 | 1.64 | 0.101 | 0046993 | .0532732 |
| landholdsz15 | .0051741 | .0045818 | 1.13 | 0.259 | 0038061 | .0141543 |
| mixedbeansha | .027079 | .1492817 | 0.18 | 0.856 | 2655078 | .3196658 |
| totharv | .0040162 | .0003832 | 10.48 | 0.000 | .0032651 | .0047673 |
| SID crops | .4112476 | .2527147 | 1.63 | 0.104 | 084064 | .9065593 |
| dinf price | .1575804 | .0931763 | 1.69 | 0.091 | 0250417 | .3402025 |
| | 0518838 | .1443977 | -0.36 | 0.719 | 3348981 | .2311304 |
| dcoop mem | .0174425 | .0888283 | 0.20 | 0.844 | 1566578 | .1915428 |
| net off farm | -4.65e-06 | 1.77e-06 | -2.63 | 0.009 | -8.11e-06 | -1.18e-06 |
| dist market | .0008576 | .0012419 | 0.69 | 0.490 | 0015764 | .0032917 |
| dadv ext | .0764455 | .0855008 | 0.89 | 0.371 | 0911329 | .244024 |
| TLU | 0033625 | .0052459 | -0.64 | 0.522 | 0136442 | .0069192 |
| AERI | 7063008 | .2357347 | -3.00 | 0.003 | -1.168332 | 2442694 |
| AERIII | .4681076 | .1154439 | 4.05 | 0.000 | .2418416 | .6943735 |
| _cons | 8983909 | .2735581 | -3.28 | 0.001 | -1.434555 | 3622269 |
| /mills | | | | | | |
| lambda | 36.19492 | 12.51386 | 2.89 | 0.004 | 11.66822 | 60.72163 |
| rho | 0.51417 | | | | | |
| sigma | 70.394405 | | | | | |

Formatted: Font: Not Bold

Formatted: Heading 2 Deleted: 3

Formatted: Font: Italic

Appendix 4. Marginal effects for the Heckman selection equation

| Average ma | arginal effects | Number of obs | = | 1,283 |
|------------|-----------------|---------------|---|-------|
| Model VCE | : Conventional | | | |

| | dy/dx | Delta-method Std. Err. | z | P> z | [95% Conf. | Interval] |
|--------------|-----------|---------------------------|-------|-------|------------|-----------|
| headage | 0012259 | .0008982 | -1.36 | 0.172 | 0029864 | .0005346 |
| dhhsex | .0126156 | .033143 | 0.38 | 0.703 | 0523434 | .0775746 |
| ae | 0004875 | .0062283 | -0.08 | 0.938 | 0126948 | .0117197 |
| edu_max | .0072268 | .0043817 | 1.65 | 0.099 | 0013611 | .0158147 |
| landholdsz15 | .0015396 | .0013607 | 1.13 | 0.258 | 0011273 | .0042065 |
| mixedbeansha | .0080576 | .0444183 | 0.18 | 0.856 | 0790006 | .0951159 |
| totharv | .0011951 | .0001052 | 11.36 | 0.000 | .0009889 | .0014012 |
| dloan | 0154386 | .0429604 | -0.36 | 0.719 | 0996394 | .0687623 |
| dcoop mem | .0051902 | .0264295 | 0.20 | 0.844 | 0466106 | .056991 |
| SID crops | .1223709 | .0749124 | 1.63 | 0.102 | 0244546 | .2691965 |
| dinf price | .0468897 | .0275942 | 1.70 | 0.089 | 0071939 | .1009733 |
| dist market | .0002552 | .0003693 | 0.69 | 0.490 | 0004686 | .000979 |
| dadv ext | .0227472 | .025419 | 0.89 | 0.371 | 0270732 | .0725675 |
| TLU | 0010005 | .0015601 | -0.64 | 0.521 | 0040583 | .0020572 |
| AERI | 210167 | .0693405 | -3.03 | 0.002 | 3460718 | 0742622 |
| AERIII | .1392902 | .033577 | 4.15 | 0.000 | .0734804 | .2050999 |
| net_off_farm | -1.38e-06 | 5.21e-07 | -2.65 | 0.008 | -2.40e-06 | -3.61e-07 |

Appendix 5. Marginal effects for the Heckman Outcome equation

. margins, dydx(*)

| Averag | ne margi | na | l effects | Number | of | obs | = | 1,283 |
|--------|----------|----|--------------|--------|----|-----|---|-------|
| Model | VCE | : | Conventional | | | | | |

| | dy/dx | Delta-method Std. Err. | d z | P> z | [95% Conf. | Interval] |
|-------------------|-----------------------|---------------------------|---------------|-------|----------------------|--------------------|
| headage dhhsex | 5501749 | .1870035 | -2.94 | 0.003 | 916695 -13.04328 | 1836548 |
| ae edu max | -4.869571 2.475904 | 1.241768 | -3.92 2.76 | 0.000 | -7.303392 .716202 | -2.435751 4.235607 |
| landholdsz15 | .1865147 | .1647481 | 1.13 | 0.258 | 1363857 | .509415 |
| mixedbeansha | 18.79244 | 4.951497 | 3.80 | 0.000 | 9.087688 | 28.4972 |
| totharv | .7886269 | .0073404 | 107.44 | 0.000 | .7742399 | .8030139 |
| dloan | 1.697414 | 8.712968 | 0.19 | 0.846 | -15.37969 | 18.77452 |
| dcoop_mem | -2.945389 | 5.388916 | -0.55 | 0.585 | -13.50747 | 7.616693 |
| SID crops | -6.173098 | 16.34557 | -0.38 | 0.706 | -38.20983 | 25.86363 |
| dinf_price | 2.042185 | 5.782466 | 0.35 | 0.724 | -9.291241 | 13.37561 |
| dist_market | .0797921 | .0717933 | 1.11 | 0.266 | 0609203 | .2205044 |
| dadv_ext | 5.772365 | 4.977633 | 1.16 | 0.246 | -3.983617 | 15.52835 |
| TLU | -1.126634 | .4213863 | -2.67 | 0.008 | -1.952536 | 3007323 |
| AERI | -37.78253 | 20.55902 | -1.84 | 0.066 | -78.07747 | 2.512406 |
| AERIII | -8.424109 | 8.036271 | -1.05 | 0.295 | -24.17491 | 7.326692 |
| net_off_farm | 0 | (omitted) | | | | |

| Appendix $\underline{6}$: | Variance | Inflation | Factor (| (VIF) | results f | for the | Multinor | nial Lo | ogit N | 4odel |
|----------------------------|----------|-----------|----------|-------|-----------|---------|----------|---------|--------|-------|
| | | | | | | | | | | |

. vif

| Variable | VIF | 1/VIF |
|---|--|--|
| mixedbeansha totharv AERIII TLU edu_max ae AERI dhhsex dcoop_mem dadv_ext headage | 1.46 1.42 1.26 1.23 1.19 1.19 1.14 1.12 1.12 1.11 1.10 | 0.686396 0.702214 0.791646 0.811692 0.837118 0.842290 0.880489 0.891081 0.894125 0.901340 0.907526 |
| net_off_farm SID_crops dloan dinf_price landholdsz15 dist_market | 1.10 1.07 1.06 1.06 1.05 1.03 | 0.907915 0.931805 0.945540 0.949392 0.970276 |
| Mean VIF | 1.16 | |

Appendix 7. Breusch-Pagan / Cook-Weisberg test for heteroscedasticity - MNL

. hettest headage dhhsex ae edu_max SID_crops landholdsz15 mixedbeansha totharv dinf
> _price dloan dcoop_mem dist_market net_off_farm dadv_ext TLU AERI AERIII

Breusch-Pagan / Cook-Weisberg test for heteroskedasticity Ho: Constant variance Variables: headage dhhsex ae edu_max SID_crops landholdsz15 mixedbeansha totharv dinf_price dloan dcoop_mem dist_market net_off_farm dadv_ext TLU AERI AERIII

chi2(17) = 18.90 Prob > chi2 = 0.3343

Appendix 8. Coefficient results of the Multinomial Logit Regression

. mlogit channel headage dhhsex ae edu_max SID_crops landholdsz15 mixedbeansha totha
> rv dinf_price dloan dcoop_mem dist_market net_off_farm dadv_ext TLU AERI AERIII, r
> obust

| Iteration | 0: | log | pseudolikelihood | = | -975.31902 |
|------------|-----|--------|------------------|---|------------|
| Iteration | 1: | log | pseudolikelihood | = | -907.99101 |
| Iteration | 2: | log | pseudolikelihood | = | -880.26845 |
| Iteration | з: | log | pseudolikelihood | = | -869.40092 |
| Iteration | 4: | log | pseudolikelihood | = | -863.66722 |
| Iteration | 5: | log | pseudolikelihood | = | -862.84222 |
| Iteration | 6: | log | pseudolikelihood | = | -862.80024 |
| Iteration | 7: | log | pseudolikelihood | = | -862.79898 |
| Iteration | 8: | log | pseudolikelihood | = | -862.79888 |
| Iteration | 9: | log | pseudolikelihood | = | -862.79885 |
| | | | | | |
| Multinomia | a 1 | logist | lc regression | | N |

 Multinomial logistic regression
 Number of obs
 =
 867

 Wald chi2(51)
 =
 924.73

 Prob > chi2
 =
 0.0000

 Log pseudolikelihood = -862.79885
 Pseudo R2
 =
 0.1154

| channel | Coef. | Robust Std. Err. | z | P> z | [95% Conf. | Interval] |
|-------------------|------------|---------------------|--------|-------|------------|-----------|
| small_scale_tra~r | (base outc | ome) | | | | |
| retailer market~r | | | | | | |
| _ headage | .0013832 | .0078057 | 0.18 | 0.859 | 0139156 | .016682 |
| dhhsex | 368568 | 2907045 | 1.27 | 0.205 | - 2012023 | 9383383 |
| ae | 032938 | .0514735 | -0.64 | 0.522 | 1338242 | .0679482 |
| edu max | .0704264 | .0373208 | 1.89 | 0.059 | 002721 | .1435737 |
| STD grops | - 2124107 | 68215 | -0.31 | 0 756 | -1 5494 | 1 124579 |
| landboldsz15 | 0046318 | 0054919 | 0.84 | 0.399 | - 0061321 | 0153958 |
| miwedbeensbe | 1963069 | 1601693 | 1 16 | 0.345 | .0001321 | 1275272 |
| hitxedbeansha | 1003908 | .1001083 | -1.10 | 0.245 | 3003209 | .12/32/3 |
| dinf price | .0001223 | 2450914 | 1.64 | 0.001 | 0003303 | .0003813 |
| dini_price | .4050024 | 2511220 | 0.25 | 0.101 | .0,04525 | .0000/0/1 |
| dagan mem | 0882999 | .3311239 | -0.23 | 0.801 | //049 | 1244013 |
| acoop_mem | 2869335 | .2098685 | -1.37 | 0.1/2 | 6982682 | .1244013 |
| dist_market | 0092983 | .003538 | -2.63 | 0.009 | 0162326 | 002364 |
| net_orr_rarm | 7.920-06 | 5.476-06 | 1.45 | 0.148 | -2.80e-06 | .0000186 |
| dadv_ext | .4459934 | .2003158 | 2.23 | 0.026 | .0533816 | .8386052 |
| TLU | 04/4245 | .0396971 | -1.19 | 0.232 | 1252293 | .0303804 |
| AERI | 3454983 | 1.276946 | -0.27 | 0./8/ | -2.848266 | 2.15/2/ |
| AERIII | 1994783 | .295029 | -0.68 | 0.499 | 7777246 | .378768 |
| | -1.596741 | .7284331 | -2.19 | 0.028 | -3.024443 | 1690381 |
| other households | | | | | | |
| headage | .0095624 | .0066183 | 1.44 | 0.149 | 0034092 | .022534 |
| dhhsex | 0876237 | .2661281 | -0.33 | 0.742 | 6092252 | .4339779 |
| ae | 0230563 | .0492188 | -0.47 | 0.639 | 1195235 | .0734108 |
| edu max | 0057781 | .0369 | -0.16 | 0.876 | 0781008 | .0665446 |
| SID crops | -1.435514 | .6134167 | -2.34 | 0.019 | -2.637788 | 2332392 |
| landholdsz15 | .0038365 | .005578 | 0.69 | 0.492 | 0070962 | .0147693 |
| mixedbeansha | -1.62701 | .50229 | -3.24 | 0.001 | -2.611481 | 64254 |
| tothary | 0054452 | 0012231 | -4.45 | 0.000 | 0078423 | 003048 |
| dinf price | 2028605 | .222444 | -0.91 | 0.362 | 6388427 | .2331217 |
| dloan | .0339579 | 3547384 | 0.10 | 0.924 | - 6613166 | 7292324 |
| dcoop mem | .1134884 | .2160661 | 0.53 | 0.599 | 3099935 | .5369703 |
| dist market | 003119 | .0027764 | -1.12 | 0.261 | 0085607 | .0023226 |
| net off farm | 0000166 | 5 570-06 | 2 99 | 0 003 | 5 720-06 | 0000275 |
| dady ext | 2289629 | 1998255 | 1 15 | 0.252 | - 1626879 | 6206137 |
| adat_che | 0110819 | 019615 | 0.60 | 0.552 | - 0254028 | 0475666 |
| A E B T | 2 512541 | 7929066 | 3 17 | 0.002 | 9596693 | 1 066413 |
| ADAT | 1593955 | 3266409 | 0.49 | 0.626 | - 4909097 | 7005009 |
| AERIII | .13333333 | .5200400 | 1 33 | 0.020 | _ 1199919 | 2 192599 |
| | .0005022 | .0004052 | 1.55 | 0.104 | .4199040 | 2.192505 |
| other_buyers | | | | | | |
| headage | .0080045 | .0130452 | 0.61 | 0.539 | 0175636 | .0335726 |
| dhhsex | .3290066 | .5105207 | 0.64 | 0.519 | 6715956 | 1.329609 |
| ae | 0639439 | .0914438 | -0.70 | 0.484 | 2431704 | .1152826 |
| edu_max | .1694168 | .068795 | 2.46 | 0.014 | .0345811 | .3042524 |
| SID_crops | -1.702307 | 1.00597 | -1.69 | 0.091 | -3.673973 | .2693579 |
| landholdsz15 | 019109 | .0162566 | -1.18 | 0.240 | 0509714 | .0127534 |
| mixedbeansha | 398667 | .2769051 | -1.44 | 0.150 | 941391 | .144057 |
| totharv | .0002786 | .0002732 | 1.02 | 0.308 | 0002567 | .000814 |
| dinf_price | 0466207 | .4552938 | -0.10 | 0.918 | 9389802 | .8457387 |
| dloan | .0781677 | .5982648 | 0.13 | 0.896 | -1.09441 | 1.250745 |
| dcoop_mem | .4615277 | .4630062 | 1.00 | 0.319 | 4459478 | 1.369003 |
| dist_market | 0069963 | .0054437 | -1.29 | 0.199 | 0176657 | .0036731 |
| net off farm | .0000184 | 5.64e-06 | 3.27 | 0.001 | 7.40e-06 | .0000295 |
| | .4736957 | .3930678 | 1.21 | 0.228 | 296703 | 1.244094 |
| TLU | 0031023 | .0230088 | -0.13 | 0.893 | 0481987 | .0419941 |
| AERI | -12.4376 | .816945 | -15.22 | 0.000 | -14.03878 | -10.83641 |
| AERIII | 0338483 | .526637 | -0.06 | 0.949 | -1.066038 | .9983411 |
| cons | -3.79329 | 1.494602 | -2.54 | 0.011 | -6.722656 | 8639239 |

| Appendix 9. Marginal Effects Results – Retailer/Marketeer Market Channel |
|--|
|--|

. mfx, predict (pr outcome(2))

| <pre>Marginal effects after mlogit y = Pr(channel==retailer_marketeer) (predict, pr outcome(2)) = .21935145</pre> | | | | | | | |
|---|----------|-----------|-------|-------|----------|---------|---------|
| variable | dy/dx | Std. Err. | Z | ₽> z | [95% | C.I.] | Х |
| headage | 0000727 | .00129 | -0.06 | 0.955 | 002598 | .002452 | 47.2964 |
| dhhsex* | .0587262 | .04074 | 1.44 | 0.149 | 02113 | .138583 | .852364 |
| ae | 0045708 | .0086 | -0.53 | 0.595 | 021418 | .012277 | 5.43706 |
| edu max | .0109857 | .00616 | 1.78 | 0.074 | 001082 | .023054 | 8.67935 |
| SID cr~s | .0137189 | .11298 | 0.12 | 0.903 | 207716 | .235153 | .561156 |
| landh~15 | .0008305 | .00088 | 0.95 | 0.343 | 000885 | .002546 | 7.35854 |
| mixedb~a | .0137793 | .02878 | 0.48 | 0.632 | 042626 | .070185 | .516103 |
| totharv | .0001622 | .00005 | 3.58 | 0.000 | .000073 | .000251 | 261.796 |
| dinf p~e* | .0700613 | .03542 | 1.98 | 0.048 | .000645 | .139477 | .770473 |
| dloan* | 0162283 | .05628 | -0.29 | 0.773 | 126539 | .094082 | .083045 |
| dcoop ~m* | 0571082 | .03718 | -1.54 | 0.125 | 129988 | .015772 | .700115 |
| dist m~t | 0014595 | .00058 | -2.52 | 0.012 | 002595 | 000324 | 26.5934 |
| net of~m | 7.84e-07 | .00000 | 0.94 | 0.347 | -8.5e-07 | 2.4e-06 | 7313.59 |
| dadv ext* | .0658352 | .03198 | 2.06 | 0.040 | .00315 | .128521 | .580161 |
| TLU | 00839 | .00681 | -1.23 | 0.218 | 021741 | .004962 | 2.3481 |
| AERI* | 149071 | .08516 | -1.75 | 0.080 | 315988 | .017846 | .014994 |
| AERIII* | 0396083 | .05261 | -0.75 | 0.452 | 142729 | .063512 | .866205 |

(*) dy/dx is for discrete change of dummy variable from 0 to 1 $\,$

Appendix 10. Marginal Effects Results – Other households Market Channel

. mfx, predict (pr outcome(3))

| <pre>Marginal effects after mlogit y = Pr(channel==other_households) (predict, pr outcome(3)) = .11996076</pre> | | | | | | | |
|---|--|--|--|---|--|---|--|
| variable | dy/dx | Std. Err. | Z | P> z | [95% | C.I.] | Х |
| headage dhhsex* ae edu_max SID_cr~s landh~15 mixedb~a totharv | .0009414 0202354 0013143 0031336 1392216 .0003588 1652814 0005792 022027 | .00068 .02936 .00504 .00373 .06428 .00053 .05147 .0001 | 1.39 -0.69 -0.26 -0.84 -2.17 0.67 -3.21 -5.86 | 0.164 0.491 0.794 0.400 0.030 0.502 0.001 0.000 | 000385 077777 011198 010436 265201 000689 266168 000773 | .002268 .037306 .008569 .004169 013243 .001406 064395 000386 | 47.2964 .852364 5.43706 8.67935 .561156 7.35854 .516103 261.796 |
| dloan* dloan* dcoop_~m* dist_m~t net_of~m dadv_ext* TLU AERI* AERIII* | 0025011 .005612 .0177299 0000569 1.47e-06 .010766 .0024301 .5300508 .0212024 | .02098 .03787 .02068 .00028 .00000 .02024 .00232 .17553 .02914 | -1.22 0.15 0.86 -0.20 2.90 0.53 1.05 3.02 0.73 | 0.222 0.882 0.391 0.841 0.004 0.595 0.294 0.003 0.467 | 068664 022803 000612 4.8e-07 028909 002108 .186011 035916 | .079827 .058262 .000498 2.5e-06 .050441 .006968 .87409 .078321 | .770473 .083045 .700115 26.5934 7313.59 .580161 2.3481 .014994 .866205 |

(*) dy/dx is for discrete change of dummy variable from 0 to 1

Appendix 11. Marginal Effects Results – Other buyers market channel

. mfx, predict (pr outcome(4))

| <pre>Marginal effects after mlogit y = Pr(channel==other_buyers) (predict, pr outcome(4)) = .03298817</pre> | | | | | | | |
|---|----------|-----------|-------|-------|---------|---------|---------|
| variable | dy/dx | Std. Err. | Z | ₽> z | [95% | C.I.] | Х |
| headage | .0002075 | .00041 | 0.51 | 0.613 | 000596 | .001011 | 47.2964 |
| dhhsex* | .0077543 | .0136 | 0.57 | 0.569 | 018906 | .034415 | .852364 |
| ae | 0017102 | .00288 | -0.59 | 0.553 | 007355 | .003934 | 5.43706 |
| edu max | .0049176 | .00198 | 2.48 | 0.013 | .001028 | .008807 | 8.67935 |
| SID cr~s | 0470858 | .0317 | -1.49 | 0.137 | 109213 | .015042 | .561156 |
| landh~15 | 0006583 | .00052 | -1.27 | 0.205 | 001676 | .000359 | 7.35854 |
| mixedb~a | 0049301 | .00884 | -0.56 | 0.577 | 022258 | .012398 | .516103 |
| totharv | .0000296 | .00001 | 3.10 | 0.002 | .000011 | .000048 | 261.796 |
| dinf p~e* | 003374 | .01508 | -0.22 | 0.823 | 032921 | .026173 | .770473 |
| dloan* | .003087 | .02021 | 0.15 | 0.879 | 036514 | .042688 | .083045 |
| dcoop ~m* | .0150582 | .01236 | 1.22 | 0.223 | 00917 | .039286 | .700115 |
| dist m~t | 0001436 | .00017 | -0.86 | 0.390 | 000471 | .000184 | 26.5934 |
| net of~m | 4.65e-07 | .00000 | 2.83 | 0.005 | 1.4e-07 | 7.9e-07 | 7313.59 |
| dadv ext* | .010776 | .01182 | 0.91 | 0.362 | 012385 | .033937 | .580161 |
| TLU | .0002003 | .00077 | 0.26 | 0.796 | 001318 | .001719 | 2.3481 |
| AERI* | 0396138 | .00782 | -5.06 | 0.000 | 054948 | 02428 | .014994 |
| AERIII* | 0001312 | .01647 | -0.01 | 0.994 | 032408 | .032145 | .866205 |

(*) dy/dx is for discrete change of dummy variable from 0 to 1 $\,$